

# Working Papers in Trade and Development

# Trends and Patterns of Foreign Direct Investments In Asia: A Comparative Perspective

Prema-chandra Athukorala

August 2009 Working Paper No. 2009/08

The Arndt-Corden Division of Economics

Research School of Pacific and Asian Studies ANU College of Asia and the Pacific

### Trends and Patterns of Foreign Direct Investments in Asia:

## A Comparative Perspective

Prema-chandra Athukorala
The Arndt-Corden Division of Economics
Research School of Pacific and Asian Studies
College of Asia and the Pacific
The Australian National University

#### **Corresponding Address:**

Prema-chandra Athukorala

The Arndt-Corden Division of Economics
Research School of Pacific & Asian Studies
College of Asia and the Pacific
The Australian National University
Canberra ACT 0200

Email: Prema-chandra.Athukorala@anu.edu.au

August 2009 Working paper No. 2009/08

Copies may be obtained from WWW Site							
http://rspas.anu.edu.au/economics/publications.php							

This Working Paper series provides a vehicle for preliminary circulation of research results in the fields of economic development and international trade. The series is intended to stimulate discussion and critical comment. Staff and visitors in

any part of the Australian National University are encouraged to contribute. To facilitate prompt distribution, papers are screened, but not formally refereed.

# Trends and Patterns of Foreign Direct Investments in Asia: A Comparative Perspective

Prema-chandra Athukorala

Arndt Corden Division of Economics Research School of Pacific and Asian Studies Australian National University

prema-chandra.athukorala@anu.edu.au

Abstract: This paper examines foreign direct investment (FDI) in developing Asia over the past three decades with emphasis on two key issues: the implications of the ongoing process of international production fragmentation and the alleged 'crowding out' effect of China's rise as a major host to FDI on the other countries in the region. The evidence suggests that assembly processes within vertically integrated global industries (in particularly, electrical goods and electronics) has gained prominence over the past two decades as the major area of attraction for foreign investors in the region. Contrary to the popular crowding out fear, China's rise as a major assembly centre within global production networks seems to have added further dynamism to region-wide MNE operations in the regions. A key policy inference from our analysis is that, in designing policies of outward-oriented development, investment and trade policies must be considered together as co-determinants of the location of production and patterns of trade.

JEL Classifications: F21, F23, O53

Key words: FDI, production fragmentation, developing Asia, China

# Trends and Patterns of Foreign Direct Investments in Asia: **An Interpretative Survey**\*

#### 1 INTRODUCTION

The purpose of this paper is to review and analyse foreign direct investment (FDI) in developing Asia over the past three decades, with emphasis on two key issues which figure prominently in the contemporary policy debate: the implications of the ongoing process of international production fragmentation for global economic integration, and the alleged 'crowding out' effects on the other countries in the region of China's rise to be the largest developing country recipient of FDI. These issues are probed against the backdrop of a comprehensive survey of emerging trends, source-country profile and industry composition of FDI flows. For the purpose of the study developing Asia is defined to cover developing East Asia (DEA), encompassing the newly industrialized economies (NIEs) in North Asia (South Korea, Taiwan and Hong Kong), China and members of the Association of Southeast Asian Nations (ASEAN), and South Asia (India, Pakistan, Bangladesh and Sri Lanka). To gain perspectives, the Asian experience is examined in the wider global context.

The paper is set out as follows. Section 2 presents an analytical account of the nature and changing patterns of FDI in developing countries. This is done in order to set the stage for the ensuing analysis. Section 3 examines overall trends in FDI and comparative performance of individual countries. Against this background, Section 4 specifically examines the implications of China's rise as the largest developing-country recipient of FDI and its implications for FDI flows to the other Asian countries. Section 5 looks at source- country composition of FDI with emphasis on trends and patterns of intra-regional flows. Section 6 deals with structural shifts in the industry profile of FDI

\* The author is greatly indebted to Shashanka Bhide for encouraging him to write this paper and the NCAER reviewer for useful suggestions for updating its literature coverage.

and the role of multinational enterprises in export expansion. The final section summarizes the key findings and draws out some general inferences.

#### 2. ANALYTICAL CONTEXT

Foreign direct investment (FDI) originates from the decision of a multinational enterprise (MNE) to relocate part of its activities in a selected host country. This decision is underpinned by the desire to reap benefits from its specific advantages (in the form of technology, managerial expertise, marketing know-how etc.), which cannot be effectively leased or purchased through 'arms length' market dealings with unrelated firms. In other words, FDI is a flow of long-term capital based on long-term profit considerations and a significant degree of influence by the investor on the management of the enterprise. It is this specific element of 'influence and control' that distinguishes FDI from portfolio investment and other forms of international capital flows (Caves 2006, Dunning 1998).

Attractiveness of a given country as a host to foreign investors is determined through a combination of its comparative advantage in international production and the domestic investment climate. The term 'investment climate' is used here in a broader sense to cover both the foreign investment regime (rules governing foreign investment and specific incentives for investors) and the general investment environment which encompasses various considerations impinging on investment decisions such as political stability, macroeconomic environment and attitudes of host countries towards foreign enterprise participation. Most economists today accept the argument that tax concessions and other profit-related incentives do not generally work unless they are appropriately combined with other initiatives to improve the general investment climate. These specific incentives are relevant for an investment decision only if the general business environment is conducive for making profit. Moreover, as countries compete for attracting investment, the incentives offered by a given country are generally counterbalanced by similar moves by other competing countries. Thus investment incentives

<sup>1</sup> According to the standard (United Nations) definition, the multinational enterprise (MNE) is a business organization that owns and controls business ventures in more than two countries, including its home country. When this definition is adopted the bulk (if not all) of FDI in a given country can be considered as MNE investment.

may matter only when other conditions are roughly similar as between alternative host countries (Wells 1986, Wells and Allen 2001, Caves 2006, Chapter 9).

Assuming a favourable investment environment, what are the specific characteristics, which determine a country's comparative advantage in international production? In answering this question, it is important to emphasize that FDI is not a homogeneous phenomenon, but complicated and finely differentiated means of globalisation of production. For the purpose of discussing factors impacting on the decision of multinational enterprises (MNEs) to locate production in a given country, it is important to distinguish between three categories of MNEs affiliates in terms of their operations in a given host country. These are, producers largely engaged in serving the domestic market ('market-seeking' investors), firms involved in extraction and processing of natural resources both for selling in the domestic market and exporting (usually for the latter purpose) ('resource seeking investors), and those engaged in production for the global market ('efficiency seeking' investors).

When it comes to market-seeking investment in developing countries, the forces explaining the location decisions of MNEs are about the same as those explaining their presence in industrialized countries. The location decision depends primarily on the prevalence in the host country of production opportunities aimed predominantly at meeting domestic demand. Given the scale economies and very small domestic markets in many developing countries, a major (if not the key) determinant of congenial domestic production is restrictions on international trade. As domestic income levels approached industrial country levels, MNEs may engage in production for serving both domestic and export markets, but MNE involvement in this area in most developing countries have so far been largely limited only to serving the domestic market, and such investments have predominantly been determined by the 'tariff jumping' motive. The so-called 'life-cycle' investors (a la Vernon 1962) who expand their production networks globally predominantly on scale-economy considerations hardly find low-income countries as attractive investment locations under free-trade conditions. In theory, under certain circumstances, MNE affiliates originally set up to serve local markets could well develop competitive advantage over the years and penetrate markets in other countries without government support (Moran 1998,). But in the real world such cases are rare and limited predominantly, if not solely, to middle-income and upper-middle-income developing countries with sizable domestic market. As Caves (2006, P. 255) aptly put it, '[G]iven scale economies and the very small domestic markets of most developing countries, a foreign subsidiary will locate their either to serve the domestic market or to export exclusively, but it will not serve the domestic market and export a little....Accordingly, generalizations that span the export and domestic market are some what suspect'

In some circumstances it may be possible to entice MNE affiliates which originally entered production to meet local markets to shift to exporting though government intervention (Bennett and Sharp 1979, Fritsch and Franco 1992, Blomström 1990). But this is typically more difficult than the encouragement of 'fresh' exportoriented investors since it requires the alteration of the firm's global production and marketing strategies. A well-known feature of MNE behaviour is that the parent company strictly controls the performance of its affiliates in the interest of global profit. The export decision of affiliates is, therefore, not just a matter of responding to domestic export incentives and government directives. Even if import-substituting MNE affiliates do respond to host Government's carrot-and-stick approach, there is no guaranty that the final outcome would justify the overall cost involved. Import-substituting production units operating in a small protected market are not usually internationally competitive. Therefore, export incentives have to be introduced and maintained at high levels to generate the anticipated export push. In addition to the related budgetary and institutional constraints, the degrees of freedom available for host countries to resort to such a interventionist policy stance is becoming increasingly limited by the ongoing efforts to enhance the contestability of global markets through international agreements on cross-border investment and competition policies under the World Trade Organization (WTO) and regional trading agreements (RTAs). On the benefit side, there may be little to gain in terms of employment generation because such exports, being simply an extension of import-substitution production, tend to be highly capital intensive (Bhagwati 2006).

For these considerations, the present-day discussion on MNE involvement in export-led industrialization in developing countries is focused almost exclusively on 'efficiency seeking' investment (commonly known as export-oriented FDI). The role of

MNEs in this sphere is 'distinctively a developing-country question' (Caves 2007, p. 257). Export-oriented FDI is, however, not a homogeneous phenomenon. Rather it is a complicated and finely differentiated means of globalisation of production. The opportunities available to a given country in mobilizing FDI in economic growth and development depend on relevant typological characteristics and the investment environment of the country and the changing pattern of international production in the global context. In order to understand the opportunities arising from the interaction of these two factors, it is important to distinguish among three different categories of exportoriented production:

- (1) Resource-based manufacturing,
- (2) Labour-intensive final consumer goods, and
- (3) Assembly processes within vertically integrated global production systems.

In the first category, the relevance for a given host country of MNE participation for export expansion depends primarily on the availability of relevant natural resources. Even if resources are available, there are other factors which may render ineffective policies designed to entice foreign investors. For instance, some processing activities, particularly those in the mineral and chemical industries, are characterized by high physical and/or human-capital intensity and may not be economical in a low-income country. A further major deterrent is cascading tariff structures in industrialized countries, which still provide heavy effective protection to domestic processing industries. Insecure property rights in resource-rich developing countries also may act as a deterrent to investors in large, capital-intensive projects. These constraints notwithstanding, there are some product areas where there are significant opportunities for successful export expansion though MNE participation. One such product line, which has gained importance over the past two decades for agricultural-resource-rich developing countries, is agro-based processed food, seafood in particular (Athukorala and Jayasuriya 2003).

For the typical developing economy, labour-intensive consumables (Category 2) are generally considered the natural starting point in the process of export-led industrialization. However, the role of MNEs in this area remains a controversial issue. In the spectacular export take-off of the East Asian NIEs in the 1960s, the key role was

played by indigenous firms with the help of marketing services provided by foreign buyers, the Japanese trading houses and the large retail buying groups in developed countries.<sup>2</sup>

There are, however, strong reasons to argue that this 'early East Asian pattern' of local-entrepreneur dominance in exports may not be replicated in latecomer countries. First, perhaps the most important factor behind the East Asian experience was the unique entrepreneurial background of these countries. Hong Kong, Taiwan and to some extent Singapore started with a stock of entrepreneurial and commercial talents inherited from the pre-revolution industrialization in China. Hong Kong and Singapore also had well established international contacts based upon entrepot trade that involved exporting manufactured goods to begin with. Likewise, the considerable industrial experience that accumulated over the preceding five decades or so under the Japanese occupation was instrumental in the export take-off in Taiwan and Korea (Amsden and Chu 2003, Rhee *et. al.* 1988). Therefore, there was no such a large difference between domestic firms in these countries and foreign firms with regard to knowledge of and access to market channels.

The present-day newcomers to export-led industrialization (including most transitional economies) are not generally comparable to the East Asian NIEs in terms of the initial level of entrepreneurial maturation. In many of these countries, the import-substitution growth strategy pursued indiscriminately over a long period has thwarted the development of local entrepreneurship. Domestic firms are generally weakly oriented towards, and have limited knowledge of, highly competitive export markets. This observation seems even more relevant for the present-day transition economies, which have embarked on the process of integration into the global economy following a long period of central planning (Lankes and Venables 1996).

Moreover, from around the mid 1980s, successful exporting firms in the East Asian NIEs have begun to play an important role as direct investors in the latecomers' labour-intensive export industries. Two main factors accounted for this trend: the erosion of international competitiveness of labour-intensive export products from their home countries as a result of rising real wages and exchange rates; and the imposition and

See Nayyar 1978, Westphal *et al.* 2002 and the work cited therein.

gradual tightening of quantitative import restrictions (QRs) under the Multifibre Arrangement (MFA) by Industrialized countries on certain labour intensive exports (mostly textile, garments and footwear) (Wells 1994). There are indications that, consistent with rapid structural transformations that are taking place in the NIEs, the intermediary role of these "new" investors in linking late comers to world markets may become increasingly important in years to come. A major advantage which investors from these new countries possess is that, unlike MNEs from developed countries they are familiar with and/or are easily adaptable to the more difficult business conditions (such as poor infrastructure, bureaucratic red tape, and unpredictable policy settings) in latecomers. Given that NIE firms have developed considerable specialized knowledge of small scale and labour-intensive production procedures in the manufacture of standardized products, they have a powerful competitive advantage over both local firms and MNEs from industrialized countries in these latecomer environments (Gereffi 1999).

The location in developing countries of relatively labour-intensive component production and assembly within vertically integrated international industries (Category 3) has been an important feature of the international division of labour since about the late 1960s. This process, which has been labeled using an array of alternative teams terms: 'global production sharing', 'international production fragmentation', vertical specialization, 'outsourcing' etc) was started by electronics MNEs based in the USA in response to increasing pressures of domestic real-wage increases and rising import competition from low cost-sources (Helleiner 1973, Krugman 1995, Feenstra 1998). The transfer abroad of component assembly operations now occurs in many industries where the technology of production permits the separation of labour-intensive components from other stages of production. Assembly operations in the electronics industry (in particular, assembly of semiconductor devices, hard disk drives and so on) are still by far the most important. The other industries with significant assembly operations located in developing countries are electrical appliances, automobile parts, electrical machinery, optical products, musical equipment, watches and cameras. In general, industries that have the potential to break up the production process to minimize the transport cost involved are more likely to move to peripheral countries than other industries.

Expansion of production fragmentation as an important facet of international production has been hastened by three mutually reinforcing developments. First, rapid advancements in production technology have enabled the industry to slice the value chain into finer, 'portable', components. Second, technological innovations in communication and transportation have shrunk the distance that once separated the world's nations, and improved the speed, efficiency and economy of coordinating geographically dispersed production processes. This has facilitated the establishment of 'services links' to combine various fragments of the production process in a timely and cost-effective manner. Third, liberalisation policy reforms in both home and host countries have considerably removed barriers to trade and investment (Jones 2000; Jones and Kierzkowski 2001). There is evidence that global trade in parts and components (middle products) is growing much faster than total manufactured exports (Athukorala and Yamashita 2009, Feenstra 1998, Yeats 2001).

At the formative stage, the process involved locating small fragments of the production process in a low-cost country and re-importing the assembled components to be incorporated in the final product. Subsequently, production networks began to encompass many countries engaged in the assembly process at different stages, resulting in multiple border crossings by product fragments before they are incorporated in the final product. Recently two other important developments in the process have set the stage for rapid expansion in the share of fragmentation-based trade in world trade. First, some fragments of the production process in certain industries have become 'standard fragments' which can be effectively used in a number of products. Second, as international networks of parts and comments supply have become firmly established, producers in advanced countries have begun to move the final assembly of an increasing range of consumer durables (for example, computers, cameras, TV sets and motor cars) to overseas locations in order to be physically closer to their final users and/or take advantage of cheap labour.

\_

<sup>&</sup>lt;sup>3</sup> Examples include long-lasting cellular batteries originally developed by computer producers and now widely used in cellular phones and electronic organizers; transmitters which are used not only in radios (as originally designed) but also in personal computers and missiles; and electronic chips, the use of which has spread beyond the computer industry into consumer electronics, motor vehicle production and many other product sectors (Brown and Linden 2005, Sturgeon 2003).

In final assembly, labour costs, while significant, are of secondary importance compared with the availability of world-class operator, technical and managerial skills; a good domestic basis of supplies and services; relatively free access to world-priced inputs including capital; and excellent infrastructure. In other words, the location decisions of MNEs in this sphere depend on the availability of a wider array of complementary inputs that enable their facilities to be efficient by world standards. Also, given the heavy initial fixed costs, MNEs are hesitant to establish overseas plants in final assembly without considerable first-hand commercial experience in the host country. For these reasons, overseas production units of MNEs involved in such final stage assembly are normally located in countries which are at a relatively advanced stage of export-led industrialization.<sup>4</sup>

MNEs from industrialized countries are the key actors in worldwide offshore assembly operations. While MNEs from the USA dominated the scene at the formative stage of global spread of assembly activities in the late 1960s, the involvement of Japanese and Western European MNEs also has been gaining importance since the late 1970s. More recently MNEs from more advanced developing countries, notably those from the East Asian NIEs, have also joined this process of internationalization of production. In response to rapid domestic wage increases, the growing reluctance of domestic labour to engage in low-paid blue-collar employment, and stringent restrictions on the importation of labour, firms in the electronics industry and other durable consumer goods industries in NIEs in East Asia have begun to produce components and sub-assemblies in neighboring countries where labour costs are still low.

Conventionally, international fragmentation of production took the form of an MNE building a subsidiary abroad to perform some of the functions that it once did at home. Thus there was a close relationship between FDI and trade in parts and components (henceforth referred to as fragmentation-based trade) within vertically integrated manufacturing industries (Helleiner 1989). However, in recent years, fragmentation practices have begun to spread beyond the domain of MNEs. As

<sup>&</sup>lt;sup>4</sup> However, as we will see below in recent years China has emerged as an important location for final assembly in many product lines largely because of the vast domestic market for these products, which naturally reduces the risk of covering the initial establishment costs (Naughton 2006, Athukorala 2009)

production operations in host countries have become firmly established, MNE subsidiaries have begun to subcontract some activities to local (host-country) firms to which they provide detailed specifications and even fragments of their own technology. At the same time, many firms which are not part of MNE networks have begun to procure components globally through arm's-length trade. Moreover, many MNEs in electronics and related industries have begun to rely increasingly on independent contract manufacturers for the operation of their global-scale production networks – a process that has been facilitated by the standardisation of some components and by advances in modular technology (Sturgeon 2003; Brown and Linden 2005). These new developments suggest that an increase in fragmentation-based trade may or may not be accompanied by an increase in the host-country stock of FDI (Brown et al. 2004: 305).

However, the bulk of fragmentation trade still takes place under the aegis of MNEs (Rangan and Lawrence 1999; Hanson et al. 2001).

In sum, the discussion in this section suggests that, in the context of emerging patterns of international division of labour, MNE involvement through FDI is bound to be more important for latecomer countries to export-led industrialization compared with the early experience of present-day NIEs. Inferences based on the early years of export-led industrialization in the East Asian NIEs may send quite inappropriate signals to policymakers in latecomer exporting countries because of the two major developments in the trade and investment environment discussed in the previous sections. First, an increasing number of firms from some NIEs have become aggressive international investors and, significantly, these 'third world' MNEs seem to possesses specific competitive advantages over 'first world' MNEs in some product areas, particularly where latecomers to export-led industrialization have a comparative advantage in international production. Second, and more importantly, the 'slicing up of the product chain' in high-tech industries, involving the cross-border reallocation of global MNE activities according to host country's relative factor endowments, has rapidly gained importance over traditional labour-intensive final goods production as the prime mover of the internationalization of production.

#### 3. Trends

Data on FDI inflows are summarised in Tables 1 and 2. Total FDI flows to developing Asia increased sharply from an average annual level of \$7 billion during 1980-04 to \$200 billion in 2006. The share of Asia in total FDI flows to developing countries increased from 29.6% to 52.6%) between these two time points. As a share of total global flows, the increase was from 9.4% to 15.3% (Table 1). FDI inflows as a share of gross domestic fixed capital (GDFCF) have been significantly higher than the comparable figure for all developing countries throughout the period 1980-1996, followed by a minor reversal in the pattern during the years of the Asian financial crisis, 1997-98. The average FDI/GDFCF ratio for developing Asia for the entire period 1970-2006 was 9.2%, compared to 6.2% for all developing countries and a global average of 7.0%.

A notable feature within developing Asia is the dramatic increase in inflows to China. Over the past two decades China has been by far the largest developing country recipient of inward FDI. For the six years 2000-06, China has been the second largest recipient of foreign investment in the world, at about \$50 billion per annum and accounting for 7% of total gross inflows, after the USA (which has received about \$140 billion per annum, or 13% of total inflows) (UNCTAD 2005). China's share in inflows to Asian developing countries increased from 11.4% during 1980-84 to 48.5% during 2000-06, and it has accounted for well over half of the total increment in FDI inflows to the region during this period.

#### Table 1 about here

#### Table 2 about here

Total FDI flows to the ASEAN countries increased sharply from an average annual level of \$3 billion in the second half of the 1980s to nearly \$30 billion during the six years before the onset of the 1997-98 financial crisis. Total FDI inflows to the region declined persistently from about \$35 billion per annum prior to 1997 to an annual average of about \$24 billion during 1997-79. However, the post-crisis experiences of individual countries vary substantially. Indonesia experienced negative FDI inflows until 2004, contributing significantly to the decline in total flows to the region. When the three atypical boom years prior to the onset of the crisis are excluded, owing to the abnormal

investor euphoria, there is no discernible break in the trend of FDI inflows to Singapore, Thailand and the Philippines. Flows to the Philippines, the country least affected by the crisis among this group, in fact continued to increase rapidly throughout. Net FDI flows to Malaysia declined from \$7.2 billion in 1996 to \$6.0 billion in 1997, a 24% contraction, and have remained virtually flat at that level from about mid-1998. This is contrast to a significant increase in flows to Korea and Thailand. It could well be that the prolonged period of policy and political uncertainty following the onset of the crisis, and widespread market skepticism about the fate of Malaysia's unorthodox reform package introduced in September 1998, may have played a role. The two extreme cases of Indonesia (continuous contraction until 2003) and the Philippine (continuous increase until its own political woes in recent years) clearly suggest the post-crisis decline in FDI inflows to the region was a temporary aberration associated with economic disruption and political turbulence caused by the crisis. Moreover, there is also evidence that the decline in FDI after the onset of the crisis was by and large limited to domestic market-oriented investment, while FDI in export-oriented industries continued to increase throughout the period, boosted by the now highly competitive exchange rates (Athukorala 2003).

It is also important to note that the continuation of the crisis-driven decline in FDI inflows to these countries well beyond the period of recovery after the crisis (that is, beyond 2000) was largely a reflection of a large overall decline in global FDI flows during 2000-2003 (UNCTAD 2005), and a global downturn in electronics. Total global FDI inflows declined from \$134 billion in 2000 to \$83 billion in 2001, \$72 billion in 2002, and \$63 billion 2003, before recovering marginally to \$65 billion 2004. Total inflows during the four years from 2001 to 2004 were 24% lower than the comparable figure for the preceding four years, 1998-2000. Interestingly, FDI inflows to the crisis-affected Asian countries (and to developing Asia in general) seemed to have been remarkably resilient in the face of this massive global contraction.

The 1990s saw a marked increase in FDI to India, a trend that represents a clear break from the preceding two decades. India's share of FDI in total developing country inflows increased from 0.4% in the 1980s to over 1.5% in the first two years of the new

<sup>&</sup>lt;sup>5</sup> This massive contraction in global FDI in an unprecedented occurrence during the entire period since 1970, when the *Word Investment Report* FDI series commenced. What caused this contradiction remains yet to be explained.

millennium. FDI as a share of GDFCF increased from less than 0.3% to over 3% between these time points. Nevertheless, the increase has to be seen in perspective. Total annual FDI inflows to India during 2000-06 amounted to a mere 10% and 8% respectively of those into China and ASEAN. A notable aspect of FDI flows to India is that they have behaved quite independently of the global trends in FDI inflows to developing countries. This pattern clearly suggests that the domestic investment climate (demand-side factors in the investment market) has been the prime mover of investment flows to the country. FDI inflows to Bangladesh, Pakistan and Sri Lanka have registered notable increases over the past two decades, but they still account for a tiny share of total flows to developing countries, and are dwarfed by those into DEA.

#### 4. THE CHINA FEAR

As we have already observed, FDI inflows to developing Asia from the mid-1990s have been dominated by inflows to China. The dramatic growth of FDI inflows to China has been accompanied by a sharp decline in the share of almost every other country in the total regional (as well as global) inflows. These contrasting patterns, coupled with some anecdotal evidence of foreign firms relocating to China (Yusuf 2003), have led to serious concern in policy circles in the region, particularly in Southeast Asia, where the growth dynamism for over two decades had relied heavily on FDI, that 'competition' from China has begun to erode their prospects for attracting FDI, hence jeopardizing a pivotal element of their outward-oriented growth strategy. Some of the FDI inflows to China could well have been at the expense of other countries, but it would be a mistake to overstate the 'China factor'.

First, there is some controversy over China's actual FDI inflows (Gunter 2004, Wee 2000, Pomfret 1989, Naughton 2006). Part of the reported FDI from Hong Kong, which has accounted for over 40% of total FDI inflows to China over the past ten years, is 'round tripping' capital. That is, it is investment that originated from the Mainland and returned to it in the guise of 'Hong Kong investment' to take advantage of tax, tariff and

<sup>&</sup>lt;sup>6</sup> The recorded increase in inflows in the past three years over the previous years partly reflects revisions to India's FDI estimation procedures, as noted above (see footnote 5).

<sup>&</sup>lt;sup>7</sup> See for instance Freeman and Bartels (2004) Chapter 1, and the work cited therein.

other benefits accorded to foreign-invested firms. The available estimates of the share of round tripping flows in total Hong Kong investment in China varies in the range of 15% to 40%. Also, the official Chinese statistics on FDI are believed to contain 'serious fat', arising from the competition among various regions and provinces to demonstrate their superior performance in attracting foreign investors. The comparison of FDI flows to China reported by the official sources with those reported by source countries in Table 3 is consistent with this view. Total investment from countries reported in the table excluding China during the years 2000-05 is almost 90% higher than the amount reported by the investing countries. Even if we make the heroic assumption that the FDI flows to Hong Kong eventually ended up in China, the difference is still significant, at about 16.2%.

#### Table 3 about here

Second, a comparison of FDI inflows to China, a relatively new host of DFI, with those to other countries with a longer history of MNE involvement, needs to be qualified for possible bias arising from the nature of the available FDI data, as reported in the *World Investment Report* and based on individual country balance of payments records. A well-known limitation of the FDI data for most countries in the region – perhaps all ASEAN countries other than Singapore and China – is that these data do not adequately capture investment financed though retained earnings. At the same time, there is convincing evidence that the relative importance of retained earnings compared to the other two components of FDI (that is, equity capital and intra-company borrowing) is positively related to the duration of MNE involvement in a given host country (Lipsey 2000). This omission is therefore likely to overstate capital inflows to China and understate those to many other countries in the region, in particular the five major ASEAN countries.

Third, investors from Hong Kong and Taiwan accounted for a large share of China's total FDI inflows, whereas over 80% of total FDI inflows to all developing countries originate from developed countries. The flows from Hong Kong and Taiwan (and also investment by ethnic Chinese investors from other countries such as Malaysia and Thailand) are presumably driven largely by ethnic links, in addition to the general

economic considerations impacting on overseas investment decisions (Huang 2003, Wee 2000, Pomfret 1989). Thus, even if the statistical errors noted above are incorporated and the official data are taken at face value, it is not realistic to assume that these flows are completely at the expense of other investment locations.

MNEs faced with the decision as to which country to invest in would naturally compare expected returns and risks across various investment locations. China may pose a particular difficulty because of the lack of well-defined property rights and the existence of political risk. Higher risk and lower expected returns may explain why some of the major source countries are not investing as much in China compared to norms based on various economic characteristics. This outcome can also explain why overseas Chinese such as those from Hong Kong and Taiwan seem to be investing a disproportionately high amount in China. In the absence of enforceable contracts, other informal instruments such as linguistic ties, family connections, geographical proximity, all of which facilitate the quicker acquisition of information, can serve as a means to increase the likelihood of securing a self-enforcing agreement (Fung 1998).

Fourthly, data on global investment patterns clearly indicate that the *measured* decline in the share in ASEAN in total developing country inflows was not entirely due to increased inflows to China. In fact, inflows to other developing countries (that is countries other than China and ASEAN) have increased at a much faster rate, from about 30% of total flows to developing countries to over 53% by 2002, compared to a *mild decline* in China's share from 32% to 28% between 1995 and 2002 (Table 1, Memorandum Items). In fact, these trends have prompted some authors to characterize China as an 'under-achiever' in attracting FDI, particularly from Europe. Much of these 'other developing country' flows were triggered by liberalization reforms in Eastern Europe, the formation of NAFTA (which triggered a massive relocation of production units from North America to Mexico) and regional cooperation initiated in other parts of Latin America.

Finally, the migration of some production processes within vertically integrated high-tech industries such as electronics, motor vehicles and cameras to China does not necessary imply a zero sum game in the competition for FDI. Rather, this process opens up opportunities for additional investment in OEM (original equipment manufacturing)

and BTO (back to office) activities in the ASEAN countries for the Chinese market. For instance, recently Intel Corporation, the world's largest computer chip maker, simultaneously invested \$200 million in a second semiconductor chip assembly and testing plant in the central Chinese city of Chengdu, in addition to its \$500 million assembly and testing facility in Shanghai. However, at the same time it invested \$40 million to expand the design and development activities in its plant in Penang, Malaysia, and also announced plans to spend \$100 million a year on further expansion of R&D activities there.8 More recently Intel signed an agreement with the government of Vietnam to set up a large electronics component assembly plant in that country, as the first step in linking Vietnam to its regional and global operational network (Athukorala and Tran 2008). The Intel story nicely fits within the broader picture of emerging patterns of manufacturing trade in the region. There is clear evidence of the rapid expansion of components and parts exports within the broader product category of machinery and transport equipment (SITC7) from the five major ASEAN countries to China (Athukorala 2009a). That is, trade in parts and components in high-tech industries is dominated by MNEs, and the FDI flows to China and other countries in the region are 'complementary' rather than 'competitive'.

#### 5. INTRA-REGIONAL FDI

There has been a significant increase in FDI outflow from countries in the region over the past three decades (Table 4). Japan emerged as a major overseas investor from the late 1960s, while for Korea, Singapore and Taiwan the outflows began to rise sharply from around the mid 1980s. The share of DEA in total global outflows is still quite small, although has been increasing rapidly, from just 0.3% in 1970-74 to over 6% in 2006. These countries, however, feature much more prominently in developing country outflows, accounting for 59% of the total in 2006, up from 41.0% in 1980-84.

#### Table 4 about here

\_

<sup>&</sup>lt;sup>8</sup> Asian Wall Street Journal, 27 August 2003. P A1 and A4

Japan's FDI in the 1980s was directed largely to North America and Europe, when these two destinations accounted for about two thirds of the total (Kawai and Urata, 1989). But the East Asian share began to increase in the 1990s, with a sharp rise in manufacturing FDI flows. The driving force was the sharp appreciation of Japanese yen during 1992-95, which substantially reduced Japan's international competitiveness. Since the mid-1980s, the geographical distribution of Japanese FDI within Asia has changed significantly, first from the NIEs to ASEAN, and then to China and other Asian countries.

As an outcome of its dramatic economic transformation over the past two decades, China itself is now becoming a significant overseas investor, predominantly in the other developing countries in the region and beyond (Chen and Lin 2007). Resource-rich countries like Indonesia, Malaysia, Laos and Cambodia have begun to attract 'resource seeking' investors from China. There is also evidence that the rapid increase in wages propelled by this fast growth has already begun to erode China's attractiveness as a low-wage investment and to entice Chinese firms involved in labour intensive manufacturing (clothing and footwear in particular) to relocate production to lower wage neighbours. For instance, Chinese investors are already the largest investors in the Cambodian garment industry and they have also begun to enter Vietnam. The imposition of punitive trade restrictions by the European Union and the USA on clothing and footwear imports from China in the mid-2005 has also driven this process.

India has a history of outward FDI dating back to the late 1950s, but total outflows remained small; total cumulative outflows up to 1990/1 amounted to a mere US\$220 million (Athukorala 2009a). Following the liberalization reforms, outflows started to increase rapidly from about the mid-1990s. In particular, there has been a real surge in outflows since about 2005 following significant dismantling of foreign exchange restrictions on capital transfers for acquisition of foreign ventures by Indian firms during 2000-04. India's share in total outward FDI of developing countries increased from below 0.5% in the early 1990s to over 6% during 2006-7 (Table 4).

How important are these intra-regional flows compared to extra-regional inflows to host countries in the region? To shed light on this issue, data on the source country

composition of FDI inflows to some Asian countries are summarised in Table 5. It is evident that, notwithstanding recent increases in intra-regional flows, the bulk of FDI inflows to Developing East Asian countries, other than to China, come from extraregional sources. However, there are significant differences among these countries in terms of relative importance of individual source countries. For instance, investors from the East Asian NIEs accounted for relatively large share of total investment in Lao PDR and Vietnam. So were investors from the EU in Lao PDR, Brunei and Myanmar (included under 'other ASEAN). A striking feature of the recent source-country profile of India compared to that of ASEAN in the relatively minor role played by investors from Japan and the East Asian NIE. This mostly reflect the fact that, despite recent reforms, the investment environment is still not conducive for efficiency seeking investment, an area where Japanese and East Asian investors generally played a more prominent role at the regional and global levels. Increase in the relative importance of investment by nonresident Indian investors (captured in 'other' sources in Table 5) has been an important feature of Indian investment approvals in recent years. China is unique for the dominance of regional investors in total inflows of FDI. During 2000-04, 52% of total FDI inflows to China originated in countries in East Asia, with Hong Kong, Korea and Taiwan accounting for 32.2%, 9.4% and 5.7% respectively. These regional flows are related to shift in production bases (mostly those involved in low-wage assembly activities to China). Thus, FDI inflow patterns in China mirror the growing importance of that country as the regional assembly center within regional production networks.

#### Table 5 about here

During the pre-reform era, over 85% of total approved FDI of Indian firms destined to other developing countries, with about a half going to other Asian countries. During the 1970s and 1980s, Indian firms, in particular the Birla Group of companies played an important role in the expansion of textile industry in Southeast Asia. Since about the mid-1990s, there has been a notable shift in the geographic profile in favour of developed-county locations. By 2007 developed countries accounted for 53% of the total approved outward FDI, up from 35% during 1991-95 (Athukorala 2009b, Table 2).

<sup>&</sup>lt;sup>9</sup> As noted above part of the reported FDI from Hong Kong is 'round tripping' capital.

#### 6. INDUSTRY PROFILE: FDI-EXPORT NEXUS

The past three decades have witnessed a profound shift, though at varying times, in the relationship between MNEs and the host countries in the region, as more and more countries have adopted an outward-oriented growth strategy. During the first two decades of the postwar period, FDI in Taiwan and Korea was predominantly involved in domestic-market oriented production. In both countries from about the mid-1960s there was a major shift in the industry composition of FDI, from the early concentration on import substitution toward export-oriented production. From about the late 1980s, FDI has played an important role in the rapid world market penetration of exports from these economies, particularly in automotive, consumer electronics and electrical goods. In Singapore, from the beginning manufacturing FDI was predominantly in 'efficiency seeking' (export oriented) production, mostly electronics. In other ASEAN countries, there has been a shift in MNE activities away from 'market seeking' (domestic-market oriented) production and towards 'efficiency-seeking' production: gradually from the mid-1970s and at an accelerated pace in the 1990s. Old-style import-substituting FDI behind tariff barriers is still found, but only in a few industries, such as automobiles and petrochemicals, and even here significant liberalizations have occurred.

Efficiency-seeking FDI in Singapore, Malaysia and the Philippines has largely concentrated in electronics. In Thailand in recent years there has been major FDI into export-oriented electronics and automotive industries; for the latter industry, the country has become the major hub for Southeast Asia. By contrast, in Indonesia efficiency-seeking FDI has continued to remain confined largely to standard labour intensive consumer goods production. Among the later-reforming countries in the region, in Vietnam, during the first decade of liberalization, FDI was heavily concentrated in domestic-market-oriented capital-intensive industries and in construction and services sectors. The period from about the late 1990s has seen a notable expansion of MNE activity into labour-intensive consumer goods production, in particular clothing, footwear and furniture. More recent years have seen some promising signs of MNE entry into

component assembly in the electronics and electrical goods industries (Athukorala and Tran 2008).<sup>10</sup>

FDI into China heavily concentrated from the beginning in export-oriented industries, more so than in Vietnam and the other transition economies. As we show in the next section, until about the mid 1990s virtually all of the industrial output of foreign-invested enterprises (FIEs) was exported. Since then the share of domestic market sales in total FIE output has gradually expanded in line with the relaxation of investment approval procedures to permit production for the vast domestic market. The share of FIEs in total exports from China has, however, expanded persistently from a mere 2% in the early 1980 to nearly 60% by 2006 (Naughton 2006). Export-oriented FIEs in China are heavily concentrated in electrical goods and electronic industries (Sun 2007).

Among major Asian economies, India still remains an outlier in process of region-wide process of increased FDI participation in export-oriented activities. In the case of India, one-third of the FDI stock at independence in 1947 was in the primary sector (plantations, mining and oil), one-quarter in manufacturing, and the rest in services, mostly trade, construction, transportation and utilities (Athreye and Kapur, 2001, Table 3). From the 1960s, inflows tended to concentrate increasingly in manufacturing, while there was also considerable divestment out of other sectors. Within manufacturing, the capital goods sector (basic metal products, machinery and transport equipment) has continued to remain the predominant recipient of FDI. Though India has an enormous supply of low-wage, low-skill manpower that could be used to attract FDI into garments and other simple assembly activities, the overall investment regime has continued to favour foreign investment in heavy industry, complex activities predominantly focused on the domestic market. There has not been any significant increase in India' penetration

1

On 28 February 2006, Intel Corporation, the world's largest semiconductor producer, announced that it will invest \$300 million (subsequently revised to 1 billion) to build a semiconductor testing and assembly plant (with an initial to absorb 1200 workers) in Ho Chi Ming City as part of its worldwide expansion of production capacity. Following Intel's arrival, the Taiwanese-based Hon Hai Precision Industry Co., the world's biggest electronics contract manufacturer announced in August 2007 its plan to set up a \$5 billion plant in Vietnam (*The Wall Street Journal*, 213 30 August 2007, p. 1). The other major players in electronics industry which have already appeared in investment approval records of the Ministry of Planning and Investment include Foxconn, Compal and Nidec (*The Wall Street Journal*, 7 October 2007, p. 1).

of world markets in industrial products in the 1990s despite the increase in FDI. The only notable exception has been the phenomenal increase in software exports since the mid 1990s (Saxenian 2002, Dosani 2007).

#### Trade-FDI Nexus

Tables 7 summarises data on the role MNE officiates in the four East Asian NIEs and seven developing Asian countries (China, Indonesia, Malaysia, the Philippines, Vietnam, India and Sri Lanka). MNE involvement in export expansion is measured here in terms of the percentage share accounted for by MNE affiliates in total manufactured exports (MNEXS) (Column 3). Export performance is measured in terms of the share of each country in total world manufactured exports (world market share, WMSH) (column 4). The final column contains summary observations on the nature of the product composition of MNE-related exports in terms of the typology developed in the previous section.

#### Table 6 about here

It is important to emphasize that these data on the MNE share in exports are pieced together from diverse sources and are therefore not strictly comparable. In particular, there is no uniform treatment of the ownership share used in identifying the 'multinationality' of host-country firms across these sources. Estimation errors in individual country figures are also unlikely to be consistent across countries, as obviously data quality varies. Nevertheless, the estimates assembled here are the best available and, taken together, they yield a number of important inferences. The twelve Asian countries covered in this table account for over ninety per cent of total manufacturing exports from the developing Asia countries (or nearly two-thirds from developing countries) over the past two decades.

Overall, there is a clear difference between the three NIEs - South Korea, Taiwan and Hong Kong – and the other countries in terms of the relationship between the share of exports accounted for my MNE affiliates (*MMEXS*) and the share in total world manufacturing exports (*WMSH*). For the former three countries, the data do not point to

any systematic relationship. By contrast, for all other countries there is a close positive relationship, suggesting that the entry of MNEs has been *export creating*. The observation that MNE involvement in export expansion from the NIEs (other than Singapore) is low by international standards generally remains valid in our data set. Nevertheless, there is evidence that FDI has played a qualitatively much more important role than that suggested by these figures. Many joint ventures in Korea, particularly those with minority ownership (which constituted almost three-quarters of all investment) were initiated by Korean entrepreneurs who approached potential foreign investors (Koo 1985, p 213). In the case of Taiwan, Ranis and Schive (1985, p 134) observe that: 'While FDI never occupied a dominant position in total manufacturing investment, it was qualitatively important in certain specific industries.'

In any case, it is important to note that in both Korea and Taiwan the MNE share in exports did increase significantly from about the mid-1970s to mid-1980s, as compared to the figures reported by Nayyar (1978) for the late 1960s. Detailed casestudies of the export performance of these countries suggest that this increase reflected the important role played by MNEs in these countries, as they shifted from the early reliance on labour intensive, standard consumer goods sectors to assembly activities in vertically integrated high-tech industries, and subsequently to sophisticated consumer durables production. 11 The available evidence on product composition of exports by MNE affiliates in Taiwan and Korea clearly attest to this important role played by these firms in the structural transformation of exports from these countries. Given the rapid expansion of traditional labour intensive exports at the initial stage of export-led growth in these countries, any analysis based on MNE shares of total exports obviously fails to capture this important point. It is interesting to note that the MNE export shares in Korea and Taiwan have tended to decline from about the mid-1980s. This is most likely due to the combined effects of exports by domestic firms growing more rapidly in recent years, and an increase in domestic sales by MNE affiliates in consumer durable industries in response to the strong growth expansion in domestic demand fuelled by rapid economic growth.

\_

<sup>&</sup>lt;sup>11</sup> Numerous studies have drawn attention to this phenomenon. See for example Koo 1985, Lee 1992, Naya 1990, Schive 1991, and Amsden and Che 2003.

The relatively small role of MNEs in export expansion from Korea and Taiwan compared to Singapore, and more importantly to the second-tier exporting countries in the region, is generally interpreted as resulting significantly from the 'guided' industrial development policies pursued by these countries. These countries, Korea in particular, so the argument goes, followed the Japanese pattern by relying on non-equity arrangements rather than FDI to access technology and other MNE-controlled assets. However, following Goh Keng Swee (1993), the architect of modern Singapore's spectacular economic development, one can argue that this difference, at least to some extent, emanated from the nature of the investment environment at the time – from the late 1960s – when technical advances in the US electronics industry began to create rapid growth of demand for semi-conductors, whose production and assembly required the intensive use of low-cost labour. At this time, China's Cultural Revolution was reaching its height, and political stability was a key factor governing the location decisions of assembly operations by electronics MNEs (Goh 1993, p. 253).

This argument receives further support from the fact that not only Korea and Taiwan which, according to the revisionists, followed 'strategic' FDI policy, but also Hong Kong, a country that followed almost laissez-faire economic policy throughout, was largely shunned by the electronics multinationals. By the time the political risk waned, and export-led growth policies became firmly rooted in these countries, wage levels had already increased to levels which made them less attractive as labour-intensive assembly locations. The electronics revolution in Singapore, which began in the mid-1960s, absorbed all unemployed labour in that country within a period of five to seven years, and electronics MNEs shifted unskilled and semi-skilled simple assembly activities to neighbouring low-wage countries — Malaysia, Thailand and Indonesia, and more recently to the Philippines. In the process, Singapore then assumed a major regional headquarters function for the electronics industry in Southeast Asia (McKendrick et al, 2002). In the following 20 years, the MNEs diversified their operations in the region, first from simple assembly to component production operations (mainly hard disc drives), and more recently to consumer electronics, such as TV sets, radios and sound systems.

The inference that MNE participation is crucial for latecomers' export success gains further support from a comparison between China and India. In China, the share of

exports from enterprises with foreign equity rose from 0.4% in 1984 to over 46% in 1996 (Table 7). This was accompanied by a more than 10-fold increase in manufactured exports over this period. By contrast, in India, where MNE subsidiaries are still predominantly of the old-fashioned 'tariff-jumping' variety, both the share of MNEs in total manufactured exports and the rate of export growth have continued to remain low. Interestingly, since the mid 1980s there has been a mild, yet persistent, decline in the MNE share of India's manufactured exports, and this decline became more pronounced following the 1991 reforms. A detailed analysis of the underlying factors is beyond the scope of this study, but the explanation seems to be in the nature of the post-reform trade and foreign investment regimes. From the early-1980 India gradually relaxed restrictions on intermediate and investment goods imports, and the removal of these restrictions was intensified as part of the liberalization reforms initiated in 1991. Consequently the pressure on MNE affiliates, which are predominantly domestic-market oriented, to export in order to become eligible for access to import (foreign exchange and quotas), gradually waned and then virtually disappeared after 1991. At the same-time, given the half-hearted nature of the policy regime relating to FDI and the still-binding bureaucratic restraints on FDI approval procedure, India has thus far not been successful in attracting exportoriented foreign investors.12

The available data do not permit precise disaggregation of exports by MNE affiliates according to the typology developed in Section 2. However, the various country case studies on the nature of the product composition of MNE-related exports (summarized in Column 5) do provide empirical support for our arguments concerning changing export patterns and the potential role of MNEs in the expansion of manufactured exports. It is evident that light manufactured goods and assembly activities within vertically integrated high-tech industries have been the main areas of MNE export activities. In Singapore, Malaysia and the Philippines, MNE involvement is predominantly in assembly activities. In the other second-tier exporting countries, the standard labour intensive products still account for the bulk of exports, but the relative importance of assembly activities seems to have increased over the years in all cases.

\_

<sup>&</sup>lt;sup>12</sup> Note that the increase in the export share in the late 1980s is consistent with the tightening of import and exchange controls in response to the balance of payments crisis preceding the 1991 liberalization.

There is also evidence of a notable shift in assembly processes, from component assembly to final good assembly in China, Thailand and Malaysia. Interestingly, there is no evidence of a shift in MNE activities from component specialization into final goods assembly in Singapore. It seems that, given the highly favorable investment climate and deep-rooted operational links coupled with relatively high domestic wages, MNEs use Singapore as the regional centre for high-tech activities in component *production*, while undertaking relatively more labour intensive assembly of components and final goods in neighboring ASEAN countries (mostly in Malaysia, Thailand and the Philippines) and China. Since the mid-1990s, China has become premier final assembly centre within global production networks of electrical goods and electronics. Affiliates of Taiwanese, Hong Kong and Korean MNEs account for the lion's share (over 80%) of assembly exports by all foreign firms located in China. However, developed-country MNEs (in particular, U.S. and Japanese MNEs) play a pivotal role in parts and component supply for these assembly firms from their home bases as well as from plants located in China and other countries in the region, in particular those located in Southeast Asia.<sup>13</sup>

Among the countries covered in the Table, Sri Lanka is unique for prolonged heavy concentration of MNE activities in standard labour intensive products (mostly garments and toys). The explanation lies in unfortunate developments in the investment climate; despite the government's continued commitment to an outward-oriented policy since the late 1970s, with further strengthening of general incentives for export-oriented FDI over the years, and the availability of cheep and trainable labour, political and policy instability has been a major deterrent to the diversification of export composition away from standard labour intensive goods to assembly activities in vertically integrated globl industries (Athukorala and Rajapatirana 2000, Chapter 6). MNEs in these industries, unlike those involved in light consumer good industries, usually view country risk and the other elements in the investment climate from a long-term perspective. Two major electronics multinationals from the USA (Motorola and Harris Corporation) had in fact finalized plans to establish large assembly plants in the Katunayake Export Processing

\_

<sup>&</sup>lt;sup>13</sup> For instance, the typical notebook computer made in a Taiwanese-owned factory in China has processing chips made by Intel in Malaysia, an operating system made by Microsoft in the U.S., a CD display screen sourced from Taiwan or Korea, and hard-disk drives sourced from Japan. Domestic value added (the cost of labour, components sourced within China, and the profit earned by foreign owned companies in China) is only one-third of the value of output (Dean and Tam 2005).

Zone in the early 1980s. These plans were abandoned as the political climate began to deteriorate. In the site selection process for MNE electronics facilities, there is something akin to a "herd psychology", particularly if the first-comer is a major player in the industry. Considering this, one can surmise that, if the Motorola and Harris projects had been completed, many other MNEs would have followed suit, giving a major boost to the expansion of assembly exports from Sri Lanka.

There is some evidence of MNE involvement in resource-based processing activities in Indonesia, Vietnam, Thailand and Sri Lanka. But the share of MNE-related exports of these product lines in total manufactured exports have declined over time in the face of rapid expansion of the standard labour intensive products and/or component assembly.

#### 7. CONCLUDING REMARKS

Over the past three decades Asia, in particular developing East Asia, has been by far the most favoured regional location for foreign direct investment in the developing world, notwithstanding a notable dip in total inflow in the aftermaths of the 1997-98 financial crisis. The rapid increase in FDI in the region has been accompanied by a notable structural shift in the composition away from traditional market-seeking (import-substitution) and towards efficiency-seeking (export-oriented) activities. Over the years, within efficiency-seeking category, FDI flows related to assembly processes within vertically integrated global industries (in particularly, electrical goods and electronics) have gained prominence over those related to traditional labour intensive manufacturing.

The across-the-bord shift in FDI towards greater export orientation, does not, however, warrant the inference that there is a 'single Asian, or even East Asia, experience' with FDI. The region is characterised by great economic diversity among countries ranging from the highly developed economies of Korea, Taiwan and Singapore to late reforming low-income countries in South Asia, and to former centrally-planned economies of Vietnam, Cambodia and Lao, which only recently reconnected to the global economy after a long period of economic isolation. There are vast differences among

these countries in the structure of their economies, and hence in their patterns of comparative advantage. This suggests that individual countries have their own niche in attracting export-oriented FDI involved in different stages of international production and production process in vertically integrated global industries.

Contrary to the popular perception, China's emergence as a major investment location is not a 'zero sum proposition' from the perspective of the region. Rather it seems to have added further dynamism to region -wide MNE operations. There are significant potential complementarity of FDI in China and other countries in the region. Migration of some production processes within vertically integrated high-tech industries such as electronics, motor vehicles and cameras to China does not necessarily imply a zero some game of competing for attracting FDI. Rather, it also opens up of opportunities for producing original-equipment-manufacturer goods and back-room operations in other countries in the region. Even if China continues to remain relatively attractive, not all stage of production within vertically integrated global industries are going to move to China. Supply chain managers are reluctant to procure all of their inputs from any one nation, preferring instead to diversify the risk of exchange rate instability or supply disruptions across countries.

India and other South Asian countries have continued to remain under performers in attracting FDI. India in particular has immense potential for becoming a major host to MNEs. It's greatest asset in this regard is a large, educated English speaking population that is willing to work at relatively law wages. In spite of widespread illiteracy, few countries can match India's combination of low-wage, highly skilled workers. The pull of a large established industrial economy like India, despite its current deficiencies and technological gaps, is also much greater than that of its smaller, less industrialized neighbours. This is not just because of the potential of its market, but because of the level of local industrial skills and experience, which could provide a fertile basis for operations of foreign firms if its liberalization process continues. In these circumstances, India could become a major destination of both market-seeking and efficiency-seeking FDI.

Despite significant since the early 1990 there are still many unresolved problems relating to the overall investment climate which make India less attractive to FDI

compared to China and other dynamic East Asian economies (UNCTAD 2002 and 2007, World Economic Forum 2008, Kearney 2007). For example, FDI is still not permitted in pure retailing (global retailers can only participate in India's retail sector trough wholesale trade or by operating retail outlets through local franchises). In apparel and other light consumer-good producing industries, which are important in export expansion and job creation at the current stage of economic development of the country, FDI is limited to 24% of total equity. Restrictions on foreign ownership of land limit the entry of foreign builders and developers in to the construction sector (See World Bank, 2003, 55-66 for details). Projects with 51% or more foreign ownership still require a long procedure of government approval. Tariff protection in India is still substantially higher than in most other developing countries, and this continues to block India's attractiveness as an export platform for labour-intensive manufacturing products. While, the 'While the License Raj' (the infamous industrial licensing policy) has been largely eliminated at the centre, it still survives at the state level, along with a pervasive 'Inspector Raj'. Private investors require a large number of permissions for gaining access to infrastructure facilities such water supply and electricity from the state governments to start business and also have to interact with the state bureaucracy in the course of day-to-day business. Stringent labour laws and other restrictive labour market practices, a weak bankruptcy framework and high corporate tax rates<sup>14</sup>, are other prominent issues. On the A.T. Kearney/Foreign Policy Magazine Globalization Index Globalization Index, an index closely-watched by the international investment community, India has continuously ranked among the bottom one percent since 2001 (when the index was first published) (Foreign Policy 2008; atkearny.com/index.php/publications/globalization-index.html).

The remarkable success in the global software and information technology industries perhaps provides a preview of India's potential to grow trough export-oriented FDI under more liberal trade and investment regimes. Software industry is the unique case in India here restrictions on MNE entry were virtually abolished. This was also accompanied by abolition of quantitative restrictions on imports of computers and peripherals and drastic cuts in import tariffs on these products. This combination of FDI

-

 $<sup>^{14}</sup>$  The corporate tax rate for foreign companies is 48% in India compared to rates in the range of 15 to 30% in East Asia.

and trade liberalization laid the foundation to make the domestic software industry internationally competitive. Now virtually every major global companies in software industry has a base in India and the entry of MNEs has opened up opportunities for Indian companies to thrive through functional specialisation, to develop niche products and services for large clients abroad. As one commentator has puts it, the success of foreign investment in the software industry is a measure of the failure of India's restrictions on foreign investment elsewhere (Desai 2002).

A key policy inference from our analysis is therefore that, in designing policies of outward-oriented development, investment and trade policies must be considered together as co-determinants of the location of production and patterns of trade. Given the fact that an increasing number of developing countries compete in attracting export-oriented FDI, countries that attempt to implement a selective FDI promotion policy are likely to lose important opportunities for export expansion. Of course, enhancing national gains from export-oriented industrialization by encouraging greater participation of local companies is a legitimate objective for any country. But under the current competitive conditions governing international production, this objective can be achieved only by providing a conducive setting for domestic entrepreneurial development as part of the overall development strategy, not through direct restrictions on the entry and operation of MNEs.

#### References

- Amsden, Alice H. and Wa-wen Chu (2003), Beyond Late Development: Taiwan's Upgrading Policies, Cambridge, MA: MIT Press.
- Athreye, Suma and Sandeep Kapur (2001), 'Private Foreign Investment in India: Pain or Panacea?', *World Economy*, 24(3), 399-424.
- Athukorala, Prema-chandra (2003), 'FDI in Crisis and Recovery: Lessons from the 1997-98 Asian Crisis', *Australian Economic History Review*, 43(2), 197-213.
- Athukorala, Prema-chandra (2006) 'Post-Crisis Export Performance: The Indonesian Experience In Regional Perspective', *Bulletin of Indonesian Economic Studies*, 42(2), 177-211.
- Athukorala, Prema-chandra (2007), *Multinational Enterprises in Asian Development*. Cheltenham: Edward Elgar.
- Athukorala, Prema-chandra (2009a) 'The Rise of China and East Asian Export Performance: Is the Crowding-out Fear Warranted?', *World Economy*, 32(2), 234–66.

- Athukorala, Prema-chandra (2009b), 'Outward Foreign Direct Investment from India', paper presented at the workshop on Outward Direct Investment from Developing Asia, 27 July, ADB Headquarters, Manila, Philippines.
- Athukorala, Prema-chandra and Jayant Menon (1996) 'Foreign Direct Investment and Industrialisation in Malaysia: Exports, Employment and Spillovers', *Asian Economic Journal*, 10(1), 29-44.
- Athukorala, Prema-chandra and Nobuaki Yamashita (2009), 'Patterns and Determinants of Production Fragmentation in World Manufacturing Trade', in Filippo di Mauro, Warwick McKibbin and Stephane Dees (eds.), *Globalization, Regionalism and Economic Interdependence*, Cambridge: Cambridge University Press, 45-72.
- Athukorala, Prema-chandra and Sarath Rajapatirana (2000), Liberalization and Industrial Transformation: Sri Lanka in International Perspective, Delhi and Oxford: Oxford University Press.
- Athukorala, Prema-chandra and Sisira Jayasuriya (2003), 'Food safety Issues, Trade and WTO Rules: A Developing Country Perspective', *World Economy*, 26(9), 613-637.
- Athukorala, Prema-chandra and Tien Quang Tran (2008), 'Foreign Direct Investment in Industrial Transition: The Experience of Vietnam', forthcoming in Prema-chandra Athukorala (ed.) *The Rise of Asia: Globalization and New Patterns of Trdae and Investment*, London: Routledge.
- Bajpai, Nirupam and Jeffrey Sachs (2000), 'Foreign Direct Investment in India: Issues and Problems', *Development Discussion Paper No. 759*, Harvard Institution for International Development.
- Bennett, Douglas and Kenneth E. Sharp (1979), 'Transnational Corporations and the Political Economy of Export promotion: The Case of the Mexican Automobile Industry', *International Organization*, 32(2), 177-201.
- Bhagwati, Jagdish (2006), 'Why Multinationals Help Reduce Poverty?', *World Economy*, 29(11), 211-228.
- Blomström, Magnus (1990), Transnational Corporations and Manufacturing Exports from Developing Countries, New York: United Nations Centre for Transnational Corporations
- Borrus, M, D Earnst and S Haggard (2000), *International Production Networks in Asia: Rivalry or Riches*? London: Rutledge.
- Brown, C. and G. Linden (2006), 'Offshoring in the semiconductor industry: a historical perspective', pp. 279–333 in L. Brainard and S.M. Collins (eds), *Brookings Trade Forum 2005: Offshoring White-Collar Work: The Issues and the Implications*, Washington DC: Brookings Institution Press.
- Brown, D.K., A.V. Deardorff and R.M. Stern (2004) 'The effect of multinational production on wages and working conditions in developing countries', pp. 279–332 in R.E. Baldwin and L.A. Winters (eds), *Challenges of Globalization: Analyzing the Economics*, Chicago: University of Chicago Press.
- Caves, Richard E. (2007), *Multinational Enterprises and Economic Analysis*, 3<sup>rd</sup> Edition, Cambridge: Cambridge University Press.

- Chen, Edward K.Y. and Ping Lin (2007), 'Emerging Transnational Corporations from East Asia: The Case of Mainland China', in Juan J. Palacios (ed.), *Multinational Corporations and the Emerging Network Economy in Asia and the Pacific*, London: Routledge, 183-203.
- Dean, Jason and Pui-Wing Tam (2005), 'The Lap Top Trail', *The Wall Street Journal*, June 9.
- Deavadasan, Evelyn S. (2007), Aspects of Labour demand and International Trade: Evidence from Malaysian Manufacturing, Kuala Lumpur: University of Malaya Press.
- Dobson, W and CS Yeu (2006). *Multinationals and East Asian integration*. Singapore: Institute of Southeast Asian Studies.
- Dosani, Rafiq (2007), 'Entrepreneurship: The True Story Behind Indian IT', in Henry S. Rowan, Marguerite G. Hancock and William F. Miller (eds.), *Making IT: The Rise of Asia in High Tech*, Sanford, CAL: Stanford University Press, 221-266.
- Desai, Ashok V. (2002), 'comment' [on Saxenian 2002] in Anne O krueger (ed.), *Economic Plocy Reforms and the Indian Economy*, Delhi and oxford: Oxford Universty Press 203-210.
- Dunning, John (1998), 'Location and the Multinational Enterprise: A Neglected factor?', *Journal of International Business Studies*, 29(1), 45-66.
- Feenstra, Robert (1998), 'Integration of Trade and Disintegration of production in the Global Economy', *Journal of economic Perspectives*, 14(4): 31-50.
- Foreign Policy (2008), 'The Globalization Index 2008', November/December.
- Freeman, NJ, and FL Bartels (eds.) (2004). *The Future of Foreign Investment in Southeast Asia*. London: Routledge.
- Fritsch, Winston and Gustavo H.B. Franco (1992), "Foreign Direct Investment and Patterns of Industrialisation and Trade in Developing Countries", in Gerald K. Helleiner (ed), *Trade Policy, Industrialisation and Development: New Perspectives*, Oxford: Clarendon Press, 197-224.
- Fung, K. C., H. Lizaka and H. Lizaka and A. Siu (2003), 'Japanese Direct Investment in Chian', *China Economic Review*, 14(3), 304-15.
- Fung, K.C. (1998), 'Accounting for Chinese Trade: Some National and Regional Considerations', in Robert E. Baldwin, Richard E. Lipsey and J David Richardson (eds.), *Geography and Ownership in Economic Accounting*, Chicago: University of Chicago Press, 173-200.
- Gereffi, Gary (1999), 'International Trade and Industrial Upgrading in the Apparel Commodity Chain', *Journal of International Economics*, 48(1), 37-70.
- Goh, Keng Swee (1993), 'What Causes Fast Economic Growth', 4<sup>th</sup> K.T. Li Lecture, 13 October 1993, Harvard University, reproduced in Linda Low (ed.), *Wealth of East Asian Nations: Speeches and Writings by Goh Keng Swee*, Singapore: Federation Press, 243-58.
- Gunter, Frank (2004), 'Capital Flights from China, 1984-2001', *China Economic Review*, 16(1), 63-85.
- Hanson, Gorden H., Raymond J. Mataloni Jr, and Mathew J. Slaughter (2001), Sasan M. Collins and Dani Rodrik (eds.), *Brookings Trade Forum 2001*, Washington DC: Brookings Institution Press, 245-282.

- Helleiner, Gerald K. (1973), 'Manufactured Exports from Less-Developed Countries and Multinational Firms', *Economic Journal*, 83(329), 21-47.
- Hill, Hal (2002), 'Industry', in Arsenio M. Balisancan and Hal Hill (eds), The Philippines Economy: Development, Policies and Challenges, New York: Oxford University Press, 219-253.
- Huang, Yanshen (2003), Selling China: Foreign Direct Investment during the Reform Era, Cambridge: Cambridge University Press.
- Jones, Ronald W. (2000), *Globalization and the Theory of Input Trade*, Cambridge, Mass.: MIT Press.
- Jones, Ronals W. & HenrykKierzkowski (2001), 'Globalization and the consequences of international fragmentation', in R. Dornbusch, G. Calvo & M. Obstfeld (Eds.), *Money, factor mobility and trade: the festschrift in honor of Robert A. Mundell*, Cambridge, Mass.: MIT Press, 365-381.
- Kohpaibook, Archanun (2006), *Multinational Enterprises in Industrial Transformation:* Evidence from Thailand, Cheltenham, UK and Northampton, USA: Edward Elgar.
- Koo, Bohn Young (1985), 'The Role of Direct Foreign Investment in Korea's Recent Economic Growth', in Walter Galenson (ed.), *Foreign Trade and Investment:* Economic Development in the Newly Industrializing Asian Countries, Madison, Wis: University of Wisconsin Press, 176-216.
- Kumar, Nagesh (2002), Globalization and the Quality of Foreign Direct Investment, Delhi: Oxford University Press.
- Krugman, Paul (1995) "Growing World Trade: Causes and Consequences", *Brookings Papers on Economic Activity*, 25th Anniversary Issue, pp. 327-77.
- Lankes, Hans-Peter and A.J. Venables (1996), 'Foreign Direct Investment in Economic Transition: The Changing Pattern of Investment', *Economics of Transition*, 4(2), 331-347.
- Lee, Joseph S. (1992), 'Capital and labour Mobility in Taiwan', in Gustav Ranis (ed.), *Taiwan: From Developing to Mature Economy*, Boulder: Westview, 305-355.
- Lipsey, Robert E. (2000), 'The Role of Foreign Direct Investment in International Capital Flows' Feldstein, Martin (ed) (2000), International Capital Flows, Chicago: University of Chicago Press, 307-330..
- McKendrick, David G., Richard F. Doner and Stephen Haggard (2000), From *Silicon Valley to Singapore: Location and Competitive Advantage in the Hard Disk Drive Industry*, Stanford, Cal: Stanford University Press.
- Moran, Theodore H. (1998), *Foreign Direct Investment and Development*, Washington DC: Institute for International Economics.
- Naughton, Barry (2007), *The Chinese Economy: Transition and Growth*, Cambridge, Mass: MIT Press.
- Nayyar, Deepak. (1978), 'Transnational Corporations and Manufactured Exports from Poor Countries', *Economic Journal*, 88, 59-84.
- Pomfret, Richard (1991), *Investing in China: Ten Years of the Open Door Policy*, New York: Harvester Wheatsheaf.
- Ramstetter, Eric D. (1997), 'Thailand: International Trade, Multinational Firms and Regional Integration' in Dobson, W. and Chia S.Y. (eds) *Multinationals and East Asian Integration*, Singapore: Institute of Southeast Asian Studies., 107 130.

- Ramstetter, Eric D. (1999), 'Comparison of Foreign Multinationls and Local Firms in Asian Manufacturing Over Time', *Asian Economic Journal*, 13(2): 163-203.
- Rangan, Subramanian and Robert Lawrence (1993), 'The Response of U.S. Firms to Exchange Rate Fluctuations: Piercing the Corporate Veil', *Brookings Papers on Economic Activity*, No 1,
- Rangan, Subramanian and Robert Z. Lawrence (1999), *A Prism on Globalization*, Washington DC: Brookings Institution Press.
- Rhee, Yung W., Bruce Ross-Larrson. and Garry Pursell (1984) *Korea's Competitive Edge: Managing the Entry into World Markets*, Baltimore: Johns Hopkins University press.
- Ryou, Jai-Won (1998), 'Korea's Outward Foreign Direct Investment and the Division of labor in the Asia Pacific', in Lee, Hiro and David W. Roland-Holst (eds.), *Economic Development and Cooperation in the Pacific Basin: Trade, Investment, Environmental Issues*, Cambridge: Cambridge University Press
- Saxenian, Annalee (2002), 'Bangalore: The Silicon Valley of Asia?', in Anne O Krueger (ed.), *Economic Policy Reforms and the Indian Economy*, Delhi and oxford: Oxford University Press, 169-193.
- Schive, Chi (1990), The Foreign Factor: The Multinational Corporation's Contribution to the Economic Modernization of the Republic of China, Standford, Cal.: Hoover Institution Press.
- Sturgeon, T.J. (2003) 'What really goes on in Silicon Valley? Spatial clustering and dispersal in modular production networks', *Journal of Economic Geography*, 3: 199–225.
- Sung, Y (2007). Made in China: From World Sweatshop to a Global Manufacturing Centre? *Asian Economic Papers* (forthcoming).
- UNCTAD (United Nations Conference on Trade and Development) (2002, 2007), World Investment Report, Geneva: United Nations.
- Vernon, Raymond (1962), 'International Investment and International Trade in the Product Cycle', *Quarterly Journal of Economics*, 80(2), 190-207.
- Wei, Shang-Jin (2000), 'Why Does China Attract so Little Foreign Direct Investment?, in Takatoshi Ito and Anne O. Krueger (eds), *The Role of Foreign Direct Investment in East Asian Economic Development*, Chicago: University of Chicago Press, 239-265.
- Wells, Luis T Jr. (1986), "Investment Incentives: An Uneasy Debate", *TNC Reporter*, 22: 17–19, 1986
- Wells, Luice T. Jr. (1994), 'Mobile Exporters: New Foreign Investors in East Asia', in Kenneth A. Froot (ed.), *Foreign Direct Investment*, Chicago: University of Chicago Press, 173-91.
- Wells, Louis T. Jr, and Nancy J. Allen (2001), 'Using Tax Incentives to Compete for Foreign Investment: Are They Worth the Cost?', *Foreign Investment Advisory Service Occasional Paper 15*, Washington DC: International Finance Corporation and World Bank, 1-67.
- Westphal, Larry E. (2002), ;Technology Strategies for Economic Development in a Fast Changing Global Economy', Economics of Innovation and New Technology, 11(4&5), 275-320.

- Wong, .Poh Kam (2007), 'The Re-Making of Singapore's High-Tech Enterprise Ecosystem', in Henry S. Rowan, Marguerite G. Hancock and William F. Miller (eds.), *Making IT: The Rise of Asia in High Tech*, Sanford, CAL: Stanford University Press, 123-174.
- World Bank (2003), *India: Sustaining Reform, Reducing Poverty*, Washington DC: World Bank.
- World Economic Forum (2008), World Competitiveness Report, Geneva: World Economic Forum.
- Yusuf, Shahid (2003), *Innovative East Asia: The Future of Growth*, New York: Oxford University Press.

**Table 1: FDI Inflows, 1944-2007** 

Table 1: FDI Inflows, 1944	1984-5 <sup>1</sup>	1989-0 <sup>1</sup>	1994-5 <sup>1</sup>	1999-0 <sup>1</sup>	2004-5 <sup>1</sup>	2006 <sup>1</sup>	2007 <sup>1</sup>
(a) US\$ millions							
World	56839	403898	597889	2486691	1676393	1411018	1833324
Developing economies <sup>2</sup>	31785	65953	219661	484805	600025	412972	499720
Economies in transition <sup>2</sup>	7	94	6163	15481	61338	57167	85942
Asia	11065	38723	144306	255550	320111	211657	248050
Japan	304	339	467	868	3037	2756	210
Developing Asia <sup>3</sup>	10761	38384	143839	254682	317074	208902	247840
East Asia	5287	17194	90350	194357	222508	131879	156706
China	3375	6880	71287	81034	133036	72715	83521
Hong Kong SDR	1021	5316	14041	86502	67650	45054	59899
Taiwan	541	2934	2934	7854	3523	7424	8161
South Korea	350	1496	2066	18887	16053	4881	2628
Southeast Asia	5189	20536	48723	52387	74336	51243	60514
Brunei Darussalam	4	17	589	1297	623	434	184
Cambodia			220	381	513	483	867
Indonesia	530	1774	6610	-6333	10233	4914	6928
Lao People's Democratic	-2	10	148	86	45	187	324
Republic							
Malaysia	1492	4279	10396	7683	8591	6048	8403
Myanmar	1	281	453	512	487	143	428
Philippines	242	1118	3050	3487	2542	2921	2928
Singapore	2349	8461	20085	33062	33758	24743	24137
Thailand	572	4412	3439	9440	13910	9010	9575
Viet Nam	1	184	3725	2773	3631	2360	6739
South Asia	285	653	4766	7937	20230	25780	30620
Bangladesh	-7	3	103	888	1306	793	666
India	125	489	3125	5753	13377	19662	22950
Maldives	1	10	16	25	24	14	15
Nepal	2	6	0	4	2	-7	6
Pakistan	103	463	1281	841	3319	4273	5333
Sri Lanka	57	61	231	374	505	480	529
(b) Share in world inflows							
Developing economies <sup>2</sup>	55.9	16.3	36.7	19.5	35.8	29.3	27.3
Economies in transition <sup>22</sup>			1.0	0.6	3.7	4.1	4.7
Asia	19.5	9.6	24.1	10.3	19.1	15.0	13.5
Japan	0.5	0.1	0.1	0.0	0.2	0.2	0.0
Developing Asia <sup>3</sup>	18.9	9.5	24.1	10.2	18.9	14.8	13.5
East Asia	9.3	4.3	15.1	7.8	13.3	9.3	8.5
China	5.9	1.7	11.9	3.3	7.9	5.2	4.6
Hong Kong SDR	1.8	1.3	2.3	3.5	4.0	3.2	3.3
Taiwan	1.0	0.7	0.5	0.3	0.2	0.5	0.4
South Korea	0.6	0.4	0.3	0.8	1.0	0.3	0.1
Southeast Asia	9.1	5.1	8.1	2.1	4.4	3.6	3.3
Indonesia	0.9	0.4	1.1	-0.3	0.6	0.3	0.4
Malaysia	2.6	1.1	1.7	0.3	0.5	0.4	0.5
Myanmar	0.0	0.1	0.1	0.0	0.0	0.0	0.0
Philippines	0.4	0.3	0.5	0.1	0.2	0.2	0.2
Singapore	4.1	2.1	3.4	1.3	2.0	1.8	1.3

Thailand	1.0	1.1	0.6	0.4	0.8	0.6	0.5
Viet Nam			0.6	0.1	0.2	0.2	0.4
South Asia	0.5	0.2	0.8	0.3	1.2	1.8	1.7
Bangladesh					0.1	0.1	
India	0.2	0.1	0.5	0.2	0.8	1.4	1.3
Pakistan	0.2	0.1	0.2	0.0	0.2	0.3	0.3
Share in inflows to developing countries							
Developing Asia <sup>3</sup>	33.9	58.2	65.5	52.5	52.8	50.6	49.6
East Asia	16.6	26.1	41.1	40.1	37.1	31.9	31.4
China	10.6	10.4	32.5	16.7	22.2	17.6	16.7
Hong Kong SDR	3.2	8.1	6.4	17.8	11.3	10.9	12.0
Taiwan	1.7	4.4	1.3	1.6	0.6	1.8	1.6
South Korea	1.1	2.3	0.9	3.9	2.7	1.2	0.5
Southeast Asia	16.3	31.1	22.2	10.8	12.4	12.4	12.1
Brunei Darussalam			0.3	0.3	0.1	0.1	0.0
Cambodia		•••	0.1	0.1	0.1	0.1	0.2
Indonesia	1.7	2.7	3.0	-1.3	1.7	1.2	1.4
Lao PDR	0.0	0.0	0.1	0.0	0.0	0.0	0.1
Malaysia	4.7	6.5	4.7	1.6	1.4	1.5	1.7
Myanmar	0.0	0.4	0.2	0.1	0.1	0.0	0.1
Philippines	0.8	1.7	1.4	0.7	0.4	0.7	0.6
Singapore	7.4	12.8	9.1	6.8	5.6	6.0	4.8
Thailand	1.8	6.7	1.6	1.9	2.3	2.2	1.9
Viet Nam	0.0	0.3	1.7	0.6	0.6	0.6	1.3
South Asia	0.9	1.0	2.2	1.6	3.4	6.2	6.1
Bangladesh				0.2	0.2	0.2	0.1
India	0.4	0.7	1.4	1.2	2.2	4.8	4.6
Pakistan	0.3	0.7	0.6	0.2	0.6	1.0	1.1
Sri Lanka	0.2	0.1	0.1	0.1	0.1	0.1	0.1

# Notes:

- 1 Annual averages
- 2 Based on the United Nations standards classification
- 3 Countries in East Asia (other than Japan), Southeast Asia (ASEAN) and South Asia.
- ... Zero or negligible.

Source: Compiled from UNCTADT World Investment database.

Table 2: FDI Inflows as % of Gross Domestic Fixed Capital Formation (GDFCF), 1984-2007

Table 2: FDI Inflows as %					ı (GDFC	(F), 1984-200
	1984-5 <sup>1</sup>	1994-5 <sup>1</sup>	1999-0 <sup>1</sup>	$2004-5^{1}$	2006	2007
World	2.2	4.8	18.3	9.0	12.9	14.8
Developed economies <sup>2</sup>	2.1	3.9	19.1	7.7	12.8	15.6
Developing economies <sup>2</sup>	2.8	8.1	15.8	11.9	12.5	12.6
	2.3	7.9	12.1	9.9	11.0	10.6
Developing Asia <sup>3</sup>						
East Asia	1.9	9.0	14.8	9.3	8.7	8.6
China	1.8	15.9	10.4	7.7	6.4	5.9
South Korea	0.6	0.6	6.6	3.8	1.9	0.9
Taiwan	2.2	2.3	5.3	2.4	9.6	10.1
South-East Asia	4.5	12.1	20.1	19.0	20.2	19.6
Brunei Darussalam	0.9	21.6	73.3	29.1	28.5	11.3
Cambodia	-	30.0	31.7	23.1	34.3	52.3
Indonesia	1.3	6.1	6.5	7.9	5.6	6.4
Lao PDR	***	38.6	25.9	2.6	17.7	26.1
Malaysia	7.4	15.2	19.2	15.9	18.5	20.6
Myanmar	0.0	23.2	31.9	18.5	7.0	20.4
Philippines	2.0	9.7	11.3	9.0	18.0	14.3
Singapore	14.3	38.6	58.4	65.7	79.9	60.0
Thailand	2.5	2.7	18.1	14.9	15.3	14.6
Viet Nam	0.0	41.5	17.5	11.1	11.6	25.4
South Asia	0.2	1.7	2.4	3.3	6.2	5.7
Bangladesh	***	0.6	4.0	4.1	4.5	3.4
Bhutan	***	***	0.3	1.4	1.2	11.3
India	0.1	1.7	2.7	3.1	6.6	5.8
Maldives	1.3	6.7	7.1	3.3	2.8	2.6
Nepal	0.2	***	0.2	0.1	***	0.4
Pakistan	0.7	4.4	3.4	8.8	16.8	17.4
Sri Lanka	1.9	3.6	4.2	4.7	6.8	6.6
NT /						

# Notes

- 1 Annual averages
- 2 Based on the United Nations standard classification
- 3 East Asia (excluding Japan) + Southeast Asia + South Asia.
- \*\*\* Data not available.

Source: Compiled from UNCTADT World Investment database

Table 3: FDI flows to China as Reported by China and by Selected Investing Countries,  $2000\text{-}05^1$ 

	As reported by China	As reported by country (US\$ n	•	Percentage difference between (1) and (2)		
	(US\$ million)	Chi	Claire a	Clair -	Claire and	
		China	China +	China	China +	
			Hong Kong		Hong Kong	
France:	3,837	2,605	4,582	47.3	-16.3	
Germany:	6,628	8,989	11,754	-26.3	-43.6	
Italy	1,526	294	322	418.4	374.5	
Japan:	28,490	18,420	22,686	54.7	25.6	
Republic of Korea:	22,267	3,570	4,124	523.7	439.9	
United Kingdom:	5,612	5,212	15,351	7.7	-63.4	
United States	25,442	11,160	21,904	128.0	16.2	
Total	93,801	50,251	80,723	86.7	16.2	

<sup>1</sup> Total for the six year period. Selection of countries was based on data availability for the entire period.

Compiled from CEIC database (China data) and OECD International Direct Investment Statistics (<a href="http://titania.sourceoecd.org">http://titania.sourceoecd.org</a>)

**Table 4: FDI Outflows, 1970-2006** 

	1984-5 <sup>1</sup>	1989-0 <sup>1</sup>	1994-5 <sup>1</sup>	1999-0 <sup>1</sup>	2004-5	2006	2007
(a) US\$ million							
World	56048	235423	324724	1159852	900480	1323150	1996514
Developing economies <sup>2</sup>	3143	15826	51270	101682	118793	212258	253145
Economies in transition <sup>2</sup>	***	***	472	2739	14242	23706	51227
Developed economies <sup>2</sup>	52905	219597	272982	1055431	767445	1087186	1692141
Asia	8472	58203	62883	87572	113045	168168	224045
Japan	6209	46077	20376	27150	38366	50266	73549
Developing Asia <sup>3</sup>	2264	12126	42507	60422	74678	117902	150496
East and North Asia	1797	10321	31053	50872	56380	82301	102865
China	382	805	2000	1345	8880	21160	22469
Hong Kong SAR	1019	2594	23219	39361	36459	44979	53187
Taiwan	76	6097	2812	5561	6587	7399	11107
South Korea	322	825	3007	4598	4478	8127	15276
South East Asia	454	1789	11340	9113	15384	22232	33466
Brunei Darussalam			55	16	44	18	38
Cambodia				8	8	8	1
Indonesia	26	11	2301	111	3237	2703	4790
Lao PDR	-1	1	3	3	***	***	***
Malaysia	226	201	2409	1724	2516	6041	10989
Philippines	37	16	200	129	384	103	3442
Singapore	165	1458	5682	6959	8873	12241	12300
Thailand	1	103	691	164	290	1032	1756
Viet Nam					33	85	150
South Asia	13	15	115	437	2915	13369	14165
Bangladesh			1	1	5	4	21
India	4	8	101	295	2579	12842	13649
Pakistan	10	14	1	16	50	109	98
Sri Lanka	1	1	7	13	22	29	95
(b) Share in global flows			150	0.0	10.0	1.50	10.5
Developing economies <sup>2</sup>	5.6	6.7	15.8	8.8	13.2	16.0	12.7
Economies in transition <sup>2</sup>			0.1	0.2	1.6	1.8	2.6
Developed economies <sup>2</sup>	94.4	93.3	84.1	91.0	85.2	82.2	84.8
Asia	15.1	24.7	19.4	7.6	12.6	12.7	11.2
Japan	11.1	19.6	6.3	2.3	4.3	3.8	3.7
Developing Asia <sup>3</sup>	4.0	5.2	13.1	5.2	8.3	8.9	7.5
East and North Asia	3.2	4.4	9.6	4.4	6.3	6.2	5.2
China	0.7	0.3	0.6	0.1	1.0	1.6	1.1
Hong Kong SAR	1.8	1.1	7.2	3.4	4.0	3.4	2.7
Taiwan	0.1	2.6	0.9	0.5	0.7	0.6	0.6
South Korea	0.6	0.4	0.9	0.4	0.5	0.6	0.8
South East Asia	0.8	0.8	3.5	0.8	1.7	1.7	1.7
Indonesia			0.7	0.0	0.4	0.2	0.2
Malaysia	0.4	0.1	0.7	0.1	0.3	0.5	0.6
Philippines	0.1		0.1	0.0	0.0	0.0	0.2
Singapore	0.3	0.6	1.7	0.6	1.0	0.9	0.6
Thailand	•••	•••	0.2	0.0	0.0	0.1	0.1

South Asia					0.3	1.0	0.7
India	•••		•••		0.3	1.0	0.7
(c ) Share in inflows to developing countries							
Developing Asia <sup>3</sup>	72.0	76.6	82.9	59.4	62.9	55.5	59.5
East and North Asia	57.2	65.2	60.6	50.0	47.5	38.8	40.6
China	12.1	5.1	3.9	1.3	7.5	10.0	8.9
Hong Kong SAR	32.4	16.4	45.3	38.7	30.7	21.2	21.0
Taiwan	2.4	38.5	5.5	5.5	5.5	3.5	4.4
South Korea	10.2	5.2	5.9	4.5	3.8	3.8	6.0
South East Asia	14.4	11.3	22.1	9.0	12.9	10.5	13.2
Indonesia	0.8	0.1	4.5	0.1	2.7	1.3	1.9
Malaysia	7.2	1.3	4.7	1.7	2.1	2.8	4.3
Philippines	1.2	0.1	0.4	0.1	0.3	0.0	1.4
Singapore	5.3	9.2	11.1	6.8	7.5	5.8	4.9
Thailand		0.7	1.3	0.2	0.2	0.5	0.7
Viet Nam							0.1
South Asia	0.4	0.1	0.2	0.4	2.5	6.3	5.6
India	0.1	0.1	0.2	0.3	2.2	6.1	5.4

### Notes:

- 4 Annual averages
- 5 Based on the United Nations standards classification
- 6 Countries in East Asia (other than Japan), Southeast Asia (ASEAN) and South Asia.
- \*\*\* Data not available
- ... Zero or negligible.

Source: Compiled from UNCTADT World Investment database.

Table 5: Source-country Composition of FDI Inflows into Selected Asian countries, 2000-05 (annual average, %)

Source country/region	on DEA North East Asia				ASEAN							South A	sia			
		Total	CHN	HK	KOR	TWN	Total	NDO	MAL	PHL	SPR	THL	VTN	Others	Total	IND
Developing East Asia (DEA)	38.6	42.0	52.3	27.8	9.5	33.8	21.5	16.5	25.6	27.4		13.0	81.2	69.4	4.8	3.0
North East Asia (NEA)	34.5	39.2	47.3	29.9	7.5	14.5	9.2	8.8	8.7	1.1		9.5	44.9	55.9	1.8	1.0
China (CHN)	6.7	7.7		26.9	6.6	1.4	0.9	3.3	0.1			1.6	4.1	9.8	0.3	
Hong Kong (HK)	18.7	21.0	32.2		0.8	9.7	6.2	2.4	7.6	0.8		6.5		41.4	0.8	
Korea (KOR)	5.5	6.3	9.4	0.9		2.7	1.7	3.1	0.3	0.1		1.4	40.8	2.0	0.7	1.0
Taiwan (TW)	3.6	4.2	5.7	2.1	0.2		0.4		0.7	0.2				2.7		
ASEAN	4.0	2.9	5.0	-2.0	2.0	19.3	12.2	7.7	16.9	26.3		3.5	36.3	13.6	3.0	2.1
Indonesia (INDO)	0.1	0.1	0.2			0.2	0.1		0.2					1.2		
Malaysia (MAL)	0.5	0.4	0.6	-0.1	0.8	2.2	1.4	5.1		0.7		1.2	24.5	4.2	0.9	
Philippines )PHL)	0.4	0.3	0.4			1.6	1.0	2.6	0.2			2.2	11.9			
Singapore (SPR)	2.8	1.9	3.6	-1.8	1.2	14.4	9.2		15.5	25.4				4.2	2.1	2.1
Thailand (THL)	0.2	0.2	0.3	-0.1		0.3	0.2		0.2	0.2				4.0		
Vietnam (VTN)	0.1					0.6	0.4		0.7							
Others																
South Asia (SA)					0.5	0.1	0.1		0.2						0.2	
India					0.4	0.1	0.1		0.1						0.2	
Japan	9.9	9.0	9.2	6.8	13.9	23.3	14.8	37.2	12.1	7.1	2.1	74.1		1.1	4.5	5.0
NAFTA	11.3	12.0	7.1	19.0	25.0	18.9	12.0		11.2	11.1	38.5	1.3	9.3	0.8	20.8	20.3
EU15	10.2	12.4	7.0	17.3	35.6	24.0	15.2	46.2	6.5	-40.6	48.6	11.6	9.5	16.1	25.2	26.9
Other Region	28.8	24.5	24.4	29.1	15.4	45.0	28.6		44.4	95.0	8.5			12.6	44.4 <sup>1</sup>	44.8 <sup>1</sup>
World	100	100	100	100	100	100	100	100	100	100	100	100	100	100	100	100

Note: --- Zero or negligible

Includes investment by non-resident Indians and local private investment 'round-tripped' via Mauritius.

Source: Compiled from UNCTAD, World Investment database.

Table 6: MNE Involvement in Manufactured Exports and Selected Export Performance Indicators in Developing Asian Countries<sup>1</sup>

		rs in Develop					
Country	Period	MNE share in exports (MNEXS) <sup>2</sup>	World market Share (WMSH) (%) <sup>3</sup>	Nature of export composition of MNE affiliates by the late 1990s <sup>4</sup> .			
(1)	(2)	(3)	(4)	(5)			
Hong Kong	1970-74	10.0*	0.52	Mostly B2 and B3.1 (Ramstetter 1999)			
	1980-84	13.8*	1.10				
	1985-89	16.0	1.19				
South	1970-74	19.3*	0.93	B3.1 and B3.2 (Nayyar 1978, Koo			
Korea	1975-79	25.0*	1.07	1985, Ramstetter 1993)			
	1980-84	25.8*	1.65				
	1985-89	26.1*	2.30				
Taiwan	1975-79	36.7	1.13	B3.1 and B3.2, with the share of the latter			
	1980-84	27.9	1.76	declining persistently since about the late			
	1990-94	19.7	2.61	1990s (Schieve 1990, Amsden and Chu 2003)			
	2000-04	10.1		,			
Singapore	1970-74	70.0	0.78	B3.1 and B3.2, with the share of the latter			
Singapore	1980-84	74.9	1.35	declining persistently since about the mid-			
	1990-94	85.2	2.16	1980s (Dobson and Chia 1996,			
	2000-04	89.1	1.52	McKendrik et al. 2000, Wong 2007)			
China	1985-89	5.3	1.49	Predominantly B2 and B3.2, with some			
	1990-94	24.3	2.44	increase in B3.1 recently (Sun 2007,			
	2000-04	53.16	9.55	Athukorala 2009)			
	2005	58.30	13.12	,			
Indonesia	1990-94	28.5	0.62	Predominantly B2, with some increase in			
	1995-99	38.5	0.67	B3.1 in recent years (Athukorala 2006)			
	2000-04	45.3*	0.68	,			
Malaysia	1975-79	65.2	0.40	Predominantly B3.1, with some (but			
	1985-89	75.6	0.59	diminishing) involvement in B3.2.			
	1990-94	78.1	1.11	(Athukorala and Menon 1996, Devadasan			
	2000-04	86.13	1.89	2006)			
	2005	87.80	1.85	,			
Philippines	1985-89	49.9*	0.16	Predominantly B3.1, with a small and			
11	1990-94	47.6*	0.21	diminishing share of B3.2.			
	2000-04	85.7*	0.07	Hill (2002)			
Thailand	1970-74	11.4*	0.15	B1, B2, B3.1 and B3.2, with shares of the			
THUHUHU	1980-84	13.5*	0.13	latter two increasing rapidly in recent			
	1700-04	15.5	0.55	Tatter two mercasing rapidly in recent			

	1990-94	50.4*	0.91	years.(Kohpaiboon 2006)
	1995-99	62.6*	1.11	
Vietnam	1990-94	12.0	0.05	Predominantly B1 (mostly seas food) and
	1995-99	39.2	0.12	B2, with a small, but an increasing, share
	2000-04	48.6	0.20	of B3.1. (Athukorala and Tran 2008)
	2004-06	56.9	0.28	
India	1970-74	5.0	0.50	A wide range of A, with some minor
	1980-84	8.7	0.40	increase in B2 and B3.1 (an IT software)
	1990-94	4.6	0.53	(Athreye 2001, Kumar 2002, Dosani
	2000-04	4.6*	0.82	2007)
Sri Lanka	1975-79	25.7*	0.01	Predominantly B2, and some B1 (mostly
	1980-84	42.8	0.03	ceramics and rubber goods) and a small
	1990-94	63.5	0.05	Percentage of B3.1
	2000-04	43.2	0.08	Athukorala and Rajapatirana (2000)
	2005	36.3	0.07	

#### Notes

- In all cases manufactured exports have been measured using the ISIC-based definition (i.e. all goods belonging to Division 3 of the International Standard Industry Classification) or an approximation to it. Figures reported are five-year averages unless otherwise indicated.
- 2 Annual averages.
- Figures marked with asterisk are for a single year or some years falling within the given five year period. For details see the Appendix.
- 4 Product categories listed in Column 7 (based on Section 2 of the paper):
  - A Exports by market-seeking MNE affiliates: product mix varies depending on the nature of import-substitution policy regime, domestic market size, export incentives and export performance requirements imposed by the government.
  - B Exports by efficiency-seeking (export-oriented) MNE affiliates.
  - B1 Resource-based manufacturing Local processing of primary products previously exported in raw state
  - B2 Standard consumer goods clothing, shoes, sporting goods.
  - B3 Assembly activities within vertically integrated production systems
  - B3.1 Parts and component assembly : : parts of electronic and electrical machinery, motor vehicle parts etc.
  - B3.2 Final assembly: computers, cameras, motor vehicles etc

Source: Athukorala (2007), Chapter 3 (updates using the same data source detailed therein)

### Working Papers in Trade and Development List of Papers (including publication details as at 2009)

- 99/1 K K TANG, 'Property Markets and Policies in an Intertemporal General Equilibrium Model'.
- 99/2 HARYO ASWICAHYONO and HAL HILL, "Perspiration' v/s 'Inspiration' in Asian Industrialization: Indonesia Before the Crisis'.
- 99/3 PETER G WARR, 'What Happened to Thailand?'.
- 99/4 DOMINIC WILSON, 'A Two-Sector Model of Trade and Growth'.
- 99/5 HAL HILL, 'Indonesia: The Strange and Sudden Death of a Tiger Economy'.
- 99/6 PREMA-CHANDRA ATHUKORALA and PETER G WARR, 'Vulnerability to a Currency Crisis: Lessons from the Asian Experience'.
- 99/7 PREMA-CHANDRA ATHUKORALA and SARATH RAJAPATIRANA, 'Liberalization and Industrial Transformation: Lessons from the Sri Lankan Experience'.
- 99/8 TUBAGUS FERIDHANUSETYAWAN, 'The Social Impact of the Indonesian Economic Crisis: What Do We Know?'
- 99/9 KELLY BIRD, 'Leading Firm Turnover in an Industrializing Economy: The Case of Indonesia'.
- 99/10 PREMA-CHANDRA ATHUKORALA, 'Agricultural Trade Liberalization in South Asia: From the Uruguay Round to the Millennium Round'.
- 99/11 ARMIDA S ALISJAHBANA, 'Does Demand for Children's Schooling Quantity and Quality in Indonesia Differ across Expenditure Classes?'
- 99/12 PREMA-CHANDRA ATHUKORALA, 'Manufactured Exports and Terms of Trade of Developing Countries: Evidence from Sri Lanka'.
- 00/01 HSIAO-CHUAN CHANG, 'Wage Differential, Trade, Productivity Growth and Education.'
- 00/02 PETER G WARR, 'Targeting Poverty.'
- 00/03 XIAOQIN FAN and PETER G WARR, 'Foreign Investment, Spillover Effects and the Technology Gap: Evidence from China.'
- 00/04 PETER G WARR, 'Macroeconomic Origins of the Korean Crisis.'

- 00/05 CHINNA A KANNAPIRAN, 'Inflation Targeting Policy in PNG: An Econometric Model Analysis.'
- 00/06 PREMA-CHANDRA ATHUKORALA, 'Capital Account Regimes, Crisis and Adjustment in Malaysia.'
- 00/07 CHANGMO AHN, 'The Monetary Policy in Australia: Inflation Targeting and Policy Reaction.'
- 00/08 PREMA-CHANDRA ATHUKORALA and HAL HILL, 'FDI and Host Country Development: The East Asian Experience.'
- 00/09 HAL HILL, 'East Timor: Development Policy Challenges for the World's Newest Nation.'
- 00/10 ADAM SZIRMAI, M P TIMMER and R VAN DER KAMP, 'Measuring Embodied Technological Change in Indonesian Textiles: The Core Machinery Approach.'
- 00/11 DAVID VINES and PETER WARR, 'Thailand's Investment-driven Boom and Crisis.'
- 01/01 RAGHBENDRA JHA and DEBA PRASAD RATH, 'On the Endogeneity of the Money Multiplier in India.'
- 01/02 RAGHBENDRA JHA and K V BHANU MURTHY, 'An Inverse Global Environmental Kuznets Curve.'
- 01/03 CHRIS MANNING, 'The East Asian Economic Crisis and Labour Migration: A Set-Back for International Economic Integration?'
- 01/04 MARDI DUNGEY and RENEE FRY, 'A Multi-Country Structural VAR Model.'
- 01/05 RAGHBENDRA JHA, 'Macroeconomics of Fiscal Policy in Developing Countries.'
- 01/06 ROBERT BREUNIG, 'Bias Correction for Inequality Measures: An application to China and Kenya.'
- 01/07 MEI WEN, 'Relocation and Agglomeration of Chinese Industry.'
- 01/08 ALEXANDRA SIDORENKO, 'Stochastic Model of Demand for Medical Care with Endogenous Labour Supply and Health Insurance.'
- 01/09 A SZIRMAI, M P TIMMER and R VAN DER KAMP, 'Measuring Embodied Technological Change in Indonesian Textiles: The Core Machinery Approach.'
- 01/10 GEORGE FANE and ROSS H MCLEOD, 'Banking Collapse and Restructuring in Indonesia, 1997-2001.'
- 01/11 HAL HILL, 'Technology and Innovation in Developing East Asia: An Interpretive Survey.'

- 01/12 PREMA-CHANDRA ATHUKORALA and KUNAL SEN, 'The Determinants of Private Saving in India.'
- 02/01 SIRIMAL ABEYRATNE, 'Economic Roots of Political Conflict: The Case of Sri Lanka.'
- 02/02 PRASANNA GAI, SIMON HAYES and HYUN SONG SHIN, 'Crisis Costs and Debtor Discipline: the efficacy of public policy in sovereign debt crises.'
- 02/03 RAGHBENDRA JHA, MANOJ PANDA and AJIT RANADE, 'An Asian Perspective on a World Environmental Organization.'
- 02/04 RAGHBENDRA JHA, 'Reducing Poverty and Inequality in India: Has Liberalization Helped?'
- 02/05 ARCHANUN KOHPAIBOON, 'Foreign Trade Regime and FDI-Growth Nexus: A Case Study of Thailand.'
- 02/06 ROSS H MCLEOD, 'Privatisation Failures in Indonesia.'
- 02/07 PREMA-CHANDRA ATHUKORALA, 'Malaysian Trade Policy and the 2001 WTO Trade Policy Review.'
- 02/08 M C BASRI and HAL HILL, 'Ideas, Interests and Oil Prices: The Political Economy of Trade Reform during Soeharto's Indonesia.'
- 02/09 RAGHBENDRA JHA, 'Innovative Sources of Development Finance Global Cooperation in the 21st Century.'
- 02/10 ROSS H MCLEOD, 'Toward Improved Monetary Policy in Indonesia.'
- 03/01 MITSUHIRO HAYASHI, 'Development of SMEs in the Indonesian Economy.'
- 03/02 PREMA-CHANDRA ATHUKORALA and SARATH RAJAPATIRANA, 'Capital Inflows and the Real Exchange Rate: A Comparative Study of Asia and Latin America.'
- 03/03 PETER G WARR, 'Industrialisation, Trade Policy and Poverty Reduction: Evidence from Asia.'
- 03/04 PREMA-CHANDRA ATHUKORALA, 'FDI in Crisis and Recovery: Lessons from the 1997-98 Asian Crisis.'
- 03/05 ROSS H McLEOD, 'Dealing with Bank System Failure: Indonesia, 1997-2002.'
- 03/06 RAGHBENDRA JHA and RAGHAV GAIHA, 'Determinants of Undernutrition in Rural India.'

- 03/07 RAGHBENDRA JHA and JOHN WHALLEY, 'Migration and Pollution.'
- 03/08 RAGHBENDRA JHA and K V BHANU MURTHY, 'A Critique of the Environmental Sustainability Index.'
- 03/09 ROBERT J BAROO and JONG-WHA LEE, 'IMF Programs: Who Is Chosen and What Are the Effects?
- 03/10 ROSS H MCLEOD, 'After Soeharto: Prospects for reform and recovery in Indonesia.'
- 03/11 ROSS H MCLEOD, 'Rethinking vulnerability to currency crises: Comments on Athukorala and Warr.'
- 03/12 ROSS H MCLEOD, 'Equilibrium is good: Comments on Athukorala and Rajapatirana.'
- 03/13 PREMA-CHANDRA ATHUKORALA and SISIRA JAYASURIYA, 'Food Safety Issues, Trade and WTO Rules: A Developing Country Perspective.'
- 03/14 WARWICK J MCKIBBIN and PETER J WILCOXEN, 'Estimates of the Costs of Kyoto-Marrakesh Versus The McKibbin-Wilcoxen Blueprint.'
- 03/15 WARWICK J MCKIBBIN and DAVID VINES, 'Changes in Equity Risk Perceptions: Global Consequences and Policy Responses.'
- 03/16 JONG-WHA LEE and WARWICK J MCKIBBIN, 'Globalization and Disease: The Case of SARS.'
- 03/17 WARWICK J MCKIBBIN and WING THYE WOO, 'The consequences of China's WTO Accession on its Neighbors.'
- 03/18 MARDI DUNGEY, RENEE FRY and VANCE L MARTIN, 'Identification of Common and Idiosyncratic Shocks in Real Equity Prices: Australia, 1982 to 2002.'
- 03/19 VIJAY JOSHI, 'Financial Globalisation, Exchange Rates and Capital Controls in Developing Countries.'
- 03/20 ROBERT BREUNIG and ALISON STEGMAN, 'Testing for Regime Switching in Singaporean Business Cycles.'
- 03/21 PREMA-CHANDRA ATHUKORALA, 'Product Fragmentation and Trade Patterns in East Asia.'
- 04/01 ROSS H MCLEOD, 'Towards Improved Monetary Policy in Indonesia: Response to De Brouwer'
- 04/02 CHRIS MANNING and PRADIP PHATNAGAR, 'The Movement of Natural Persons in Southeast Asia: How Natural?

- 04/03 RAGHBENDRA JHA and K V BHANU MURTHY, 'A Consumption Based Human Development Index and The Global Environment Kuznets Curve'
- 04/04 PREMA-CHANDRA ATHUKORALA and SUPHAT SUPHACHALASAI, 'Post-crisis Export Performance in Thailand'
- 04/05 GEORGE FANE and MARTIN RICHARDSON, 'Capital gains, negative gearing and effective tax rates on income from rented houses in Australia'
- 04/06 PREMA-CHANDRA ATHUKORALA, 'Agricultural trade reforms in the Doha Round: a developing country perspective'
- 04/07 BAMBANG-HERU SANTOSA and HEATH McMICHAEL, 'Industrial development in East Java: A special case?'
- 04/08 CHRIS MANNING, 'Legislating for Labour Protection: Betting on the Weak or the Strong?'
- 05/01 RAGHBENDRA JHA, 'Alleviating Environmental Degradation in the Asia-Pacific Region: International cooperation and the role of issue-linkage'
- 05/02 RAGHBENDRA JHA, RAGHAV GAIHA and ANURAG SHARMA, 'Poverty Nutrition Trap in Rural India'
- 05/03 PETER WARR, 'Food Policy and Poverty in Indonesia: A General Equilibrium Analysis'
- 05/04 PETER WARR, 'Roads and Poverty in Rural Laos'
- 05/05 PREMA-CHANDRA ATHUKORALA and BUDY P RESOSUDARMO, 'The Indian Ocean Tsunami: Economic Impact, Disaster Management and Lessons'
- 05/06 PREMA-CHANDRA ATHUKORALA, 'Trade Policy Reforms and the Structure of Protection in Vietnam'
- 05/07 PREMA-CHANDRA ATHUKORALA and NOBUAKI YAMASHITA, 'Production Fragmentation and Trade Integration: East Asia in a Global Context'
- 05/08 ROSS H MCLEOD, 'Indonesia's New Deposit Guarantee Law'
- 05/09 KELLY BIRD and CHRIS MANNING, 'Minimum Wages and Poverty in a Developing Country: Simulations from Indonesia's Household Survey'
- 05/10 HAL HILL, 'The Malaysian Economy: Past Successes, Future Challenges'
- 05/11 ZAHARI ZEN, COLIN BARLOW and RIA GONDOWARSITO, 'Oil Palm in Indonesian Socio-Economic Improvement: A Review of Options'

- 05/12 MEI WEN, 'Foreign Direct Investment, Regional Geographical and Market Conditions, and Regional Development: A Panel Study on China'
- 06/01 JUTHATHIP JONGWANICH, 'Exchange Rate Regimes, Capital Account Opening and Real Exchange Rates: Evidence from Thailand'
- 06/02 ROSS H MCLEOD, 'Private Sector Lessons for Public Sector Reform in Indonesia'
- 06/03 PETER WARR, 'The Gregory Thesis Visits the Tropics'
- 06/04 MATT BENGE and GEORGE FANE, 'Adjustment Costs and the Neutrality of Income Taxes'
- 06/05 RAGHBENDRA JHA, 'Vulnerability and Natural Disasters in Fiji, Papua New Guinea, Vanuatu and the Kyrgyz Republic'
- 06/06 PREMA-CHANDRA ATHUKORALA and ARCHANUN KOHPAIBOON, 'Multinational Enterprises and Globalization of R&D: A Study of U.S-based Firms
- 06/07 SANTANU GUPTA and RAGHBENDRA JHA, 'Local Public Goods in a Democracy: Theory and Evidence from Rural India'
- 06/08 CHRIS MANNING and ALEXANDRA SIDORENKO, 'The Regulation of Professional Migration in ASEAN Insights from the Health and IT Sectors'
- 06/09 PREMA-CHANDRA ATHUKORALA, 'Multinational Production Networks and the New Geo-economic Division of Labour in the Pacific Rim'
- 06/10 RAGHBENDRA JHA, RAGHAV GAIHA and ANURAG SHARMA, 'On Modelling Variety in Consumption Expenditure on Food'
- 06/11 PREMA-CHANDRA ATHUKORALA, 'Singapore and ASEAN in the New Regional Division of Labour'
- 06/12 ROSS H MCLEOD, 'Doing Business in Indonesia: Legal and Bureaucratic Constraints'
- 06/13 DIONISIUS NARJOKO and HAL HILL, 'Winners and Losers during a Deep Economic Crisis; Firm-level Evidence from Indonesian Manufacturing'
- 06/14 ARSENIO M BALISACAN, HAL HILL and SHARON FAYE A PIZA, 'Regional Development Dynamics and Decentralization in the Philippines: Ten Lessons from a 'Fast Starter'
- 07/01 KELLY BIRD, SANDY CUTHBERTSON and HAL HILL, 'Making Trade Policy in a New Democracy after a Deep Crisis: Indonesia
- 07/02 RAGHBENDRA JHA and T PALANIVEL, 'Resource Augmentation for Meeting the Millennium Development Goals in the Asia Pacific Region'

- 07/03 SATOSHI YAMAZAKI and BUDY P RESOSUDARMO, 'Does Sending Farmers Back to School have an Impact? A Spatial Econometric Approach'
- 07/04 PIERRE VAN DER ENG, 'De-industrialisation' and Colonial Rule: The Cotton Textile Industry in Indonesia, 1820-1941'
- 07/05 DJONI HARTONO and BUDY P RESOSUDARMO, 'The Economy-wide Impact of Controlling Energy Consumption in Indonesia: An Analysis Using a Social Accounting Matrix Framework'
- 07/06 W MAX CORDEN, 'The Asian Crisis: A Perspective after Ten Years'
- 07/07 PREMA-CHANDRA ATHUKORALA, 'The Malaysian Capital Controls: A Success Story?
- 07/08 PREMA-CHANDRA ATHUKORALA and SATISH CHAND, 'Tariff-Growth Nexus in the Australian Economy, 1870-2002: Is there a Paradox?,
- 07/09 ROD TYERS and IAN BAIN, 'Appreciating the Renbimbi'
- 07/10 PREMA-CHANDRA ATHUKORALA, 'The Rise of China and East Asian Export Performance: Is the Crowding-out Fear Warranted?
- 08/01 RAGHBENDRA JHA, RAGHAV GAIHA AND SHYLASHRI SHANKAR, 'National Rural Employment Guarantee Programme in India A Review'
- 08/02 HAL HILL, BUDY RESOSUDARMO and YOGI VIDYATTAMA, 'Indonesia's Changing Economic Geography'
- 08/03 ROSS H McLEOD, 'The Soeharto Era: From Beginning to End'
- 08/04 PREMA-CHANDRA ATHUKORALA, 'China's Integration into Global Production Networks and its Implications for Export-led Growth Strategy in Other Countries in the Region'
- 08/05 RAGHBENDRA JHA, RAGHAV GAIHA and SHYLASHRI SHANKAR, 'National Rural Employment Guarantee Programme in Andhra Pradesh: Some Recent Evidence'
- 08/06 NOBUAKI YAMASHITA, 'The Impact of Production Fragmentation on Skill Upgrading: New Evidence from Japanese Manufacturing'
- 08/07 RAGHBENDRA JHA, TU DANG and KRISHNA LAL SHARMA, 'Vulnerability to Poverty in Fiji'
- 08/08 RAGHBENDRA JHA, TU DANG, 'Vulnerability to Poverty in Papua New Guinea'

- 08/09 RAGHBENDRA JHA, TU DANG and YUSUF TASHRIFOV, 'Economic Vulnerability and Poverty in Tajikistan'
- 08/10 RAGHBENDRA JHA and TU DANG, 'Vulnerability to Poverty in Select Central Asian Countries'
- 08/11 RAGHBENDRA JHA and TU DANG, 'Vulnerability and Poverty in Timor-Leste'
- 08/12 SAMBIT BHATTACHARYYA, STEVE DOWRICK and JANE GOLLEY, 'Institutions and Trade: Competitors or Complements in Economic Development?
- 08/13 SAMBIT BHATTACHARYYA, 'Trade Liberalizaton and Institutional Development'
- 08/14 SAMBIT BHATTACHARYYA, 'Unbundled Institutions, Human Capital and Growth'
- 08/15 SAMBIT BHATTACHARYYA, 'Institutions, Diseases and Economic Progress: A Unified Framework'
- 08/16 SAMBIT BHATTACHARYYA, 'Root causes of African Underdevelopment'
- 08/17 KELLY BIRD and HAL HILL, 'Philippine Economic Development: A Turning Point?'
- 08/18 HARYO ASWICAHYONO, DIONISIUS NARJOKO and HAL HILL, 'Industrialization after a Deep Economic Crisis: Indonesia'
- 08/19 PETER WARR, 'Poverty Reduction through Long-term Growth: The Thai Experience'
- 08/20 PIERRE VAN DER ENG, 'Labour-Intensive Industrialisation in Indonesia, 1930-1975: Output Trends and Government policies'
- 08/21 BUDY P RESOSUDARMO, CATUR SUGIYANTO and ARI KUNCORO, 'Livelihood Recovery after Natural Disasters and the Role of Aid: The Case of the 2006 Yogyakarta Earthquake'
- 08/22 PREMA-CHANDRA ATHUKORALA and NOBUAKI YAMASHITA, 'Global Production Sharing and US-China Trade Relations'
- 09/01 PIERRE VAN DER ENG, 'Total Factor Productivity and the Economic Growth in Indonesia'
- 09/02 SAMBIT BHATTACHARYYA and JEFFREY G WILLIAMSON, 'Commodity Price Shocks and the Australian Economy since Federation'
- 09/03 RUSSELL THOMSON, 'Tax Policy and the Globalisation of R & D'
- 09/04 PREMA-CHANDRA ATHUKORALA, 'China's Impact on Foreign Trade and Investment in other Asian Countries'

- 09/05 PREMA-CHANDRA ATHUKORALA, 'Transition to a Market Economy and Export Performance in Vietnam'
- 09/06 DAVID STERN, 'Interfuel Substitution: A Meta-Analysis'
- 09/07 PREMA-CHANDRA ATHUKORALA and ARCHANUN KOHPAIBOON, 'Globalization of R&D US-based Multinational Enterprises'
- 09/08 PREMA-CHANDRA ATHUKORALA, 'Trends and Patterns of Foreign Investments in Asia: A Comparative Perspective'